SECURITIES AND EXCHANGE COMMISSION

FORM 6-K

Current report of foreign issuer pursuant to Rules 13a-16 and 15d-16 Amendments

Filing Date: 2021-06-25 | Period of Report: 2021-06-25 SEC Accession No. 0001564590-21-034609

(HTML Version on secdatabase.com)

FILER

Liminal BioSciences Inc.

CIK:1351172| IRS No.: 000000000 | State of Incorp.:Z4 | Fiscal Year End: 1231

Type: 6-K | Act: 34 | File No.: 001-39131 | Film No.: 211048431

SIC: 2834 Pharmaceutical preparations

Mailing Address FRAPPIER #300 LAVAL A8 H7V 4B4

Business Address 440 BOULEVARD ARMAND 440 BOULEVARD ARMAND FRAPPIER #300 LAVAL A8 H7V 4B4 450-781-0115

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 6-K

REPORT OF FOREIGN PRIVATE ISSUER PURSUANT TO RULE 13a-16 OR 15d-16 UNDER THE SECURITIES EXCHANGE ACT OF 1934

For the Month of June 2021

Commission File Number: 001-39131

LIMINAL BIOSCIENCES INC.

(Translation of registrant's name into English)

440 Armand-Frappier Boulevard, Suite 300 Laval, Québec H7V 4B4 (Address of principal executive office)

Indicate by check mark whether the registrant files or will file annual reports under cover of Form 20-F or Form 40-F: ☑ Form 20-F □ Form 40-F
Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(1): □
Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(7): □

INCORPORATION BY REFERENCE

This Report on Form 6-K (the "Report") and Exhibits 99.1 and 99.2 to this Report is hereby expressly incorporated by reference into the registrant's registration statements on Form F-3 (File nos. 333-251055, 333-245703 and 333-251065) filed with the Securities and Exchange Commission on December 1, 2020, December 2, 2020 and December 2, 2020, respectively, and the registration statement on Form S-8 (File no. 333-235692) filed with the Securities and Exchange Commission on December 23, 2019.

EXHIBIT LIST

Exhibit	Description
99.1	Material Change Report dated June 25, 2021
99.2#	Share Purchase Agreement between Liminal BioSciences Inc. and Kedrion S.p.A. dated June 22, 2021
#	Certain portions of this exhibit (indicated by asterisks) have been omitted because they are not material.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Liminal BioSciences Inc.

Date: June 25, 2021 By: /s/ Bruce Pritchard

Name Bruce Pritchard

Title: Chief Executive Officer

FORM 51 102F3 MATERIAL CHANGE REPORT

Item 1 Name and Address of Company

Liminal BioSciences Inc. (the "Company" or "Liminal") 440 Armand-Frappier Blvd., Suite 300 Laval, Québec, H7V 4B4

Item 2 Date of Material Change

June 22, 2021.

Item 3 News Release

The press release attached as Schedule "A" was disseminated through Cision Canada on June 23, 2021 with respect to the material change and was filed on SEDAR.

Item 4 Summary of Material Change

Liminal Biosciences announces signature of a share purchase agreement for sale of remaining plasma-derived business with Kedrion S.p.A. ("Kedrion").

Item 5 Full Description of Material Change

The Company announced today that, Kedrion S.p.A ("Kedrion") has exercised its option to acquire the remainder of the Company's plasma-derived business under the terms of the Share Purchase Agreement entered into by the Company and Kedrion. This acquisition would include the Ryplazim® (plasminogen, human-tvmh) ("Ryplazim®") business operated through its subsidiaries, Prometic Bioproduction Inc., the Company's plasma-derived therapeutics manufacturing facility ("PBP"), and Prometic Biotherapeutics Inc. ("PBT"), holder of the biological license application ("BLA") for Ryplazim® and associated Rare Pediatric Disease Priority Review Voucher ("PRV"). On June 4, 2021, Ryplazim was approved by the U.S. Food and Drug Administration (FDA) as the first treatment of patients with plasminogen deficiency type 1 (hypoplasminogenia).

The Share Purchase Agreement provides for payment of USD 5 million payable upon closing of the sale of the shares of PBP (the "First Closing") and provides for the right for Liminal BioSciences to receive from PBT an amount equal to seventy percent (70%) of the net proceeds obtained from the sale of a PRV prior the closing of the sale of the shares of PBT (the "Second Closing"). Upon closing of the two transactions, Kedrion would assume all development, manufacturing and commercialization activities and operating costs for Ryplazim®.

The closing of the Transactions are subject to the fulfilment of the certain customary conditions precedent.

Item 6 Reliance on subsection 7.1(2) of National Instrument 51-102

Not applicable.

Item 7 Omitted Information

Not applicable.

Item 8 Executive Officer

The foregoing accurately discloses the material change referred to in this report and inquiries in respect of the material change referred to in this report may be made to:

Marie Iskra General Counsel +1 (438) 807-1397 m.iskra@liminalbiosciences.com

Item 9 Date of Report

June 25, 2021.



Schedule "A"

Press Release For immediate release

Liminal Biosciences Announces Signature of a Share Purchase Agreement for sale of Remaining Plasma-derived Business with Kedrion

LAVAL, CANADA, and CAMBRIDGE, ENGLAND –June 23, 2021 – Liminal BioSciences Inc. (Nasdaq: LMNL) ("Liminal BioSciences" or the "Company"), announced today that, Kedrion S.p.A ("Kedrion") has exercised its option to acquire the remainder of the Company's plasma-derived business under the terms of the Share Purchase Agreement entered into by the Company and Kedrion. This acquisition would include the Ryplazim® (plasminogen, human-tvmh) ("Ryplazim®") business operated through its subsidiaries, Prometic Bioproduction Inc., the Company's plasma-derived therapeutics manufacturing facility ("PBP"), and Prometic Biotherapeutics Inc. ("PBT"), holder of the biological license application ("BLA") for Ryplazim® and associated Rare Pediatric Disease Priority Review Voucher ("PRV"). On June 4, 2021, Ryplazim was approved by the U.S. Food and Drug Administration (FDA) as the first treatment of patients with plasminogen deficiency type 1 (hypoplasminogenia).

The Share Purchase Agreement provides for payment of USD 5 million payable upon closing of the sale of the shares of PBP (the "First Closing") and provides for the right for Liminal BioSciences to receive from PBT an amount equal to seventy percent (70%) of the net proceeds obtained from the sale of a PRV prior the closing of the sale of the shares of PBT (the "Second Closing"). Upon closing of the two transactions, Kedrion would assume all development, manufacturing and commercialization activities and operating costs for Ryplazim®.

"The execution of the Share Purchase Agreement is a key step in our strategy to refocus our resources on the development of our small molecule pipeline", said Bruce Pritchard, CEO of Liminal BioSciences. "This transaction also reflects Liminal

Press Release for immediate release



BioSciences' commitment to delivering innovative medicines to patients either by ourselves or through strategic partnerships."

"With extensive experience in the plasma market, Kedrion has the experience and established network to rapidly drive the development and commercialization of Ryplazim® for patients with plasminogen deficiency type 1(hypoplasminogenia)," stated Patrick Sartore, President of Liminal BioSciences.

The closing of the Transactions are subject to the fulfilment of the certain customary conditions precedent.

About Liminal BioSciences Inc.

Liminal BioSciences is a biopharmaceutical company focused on discovering, developing and commercializing novel treatments for patients suffering from diseases of unmet medical need, primarily related to fibrosis, including respiratory, liver and kidney diseases. In December 2020, Liminal BioSciences' lead small molecule product candidate, fezagepras (PBI-4050), entered a Phase 1 clinical trial in in the UK to evaluate multiple-ascending doses in normal healthy volunteers, at daily dose exposures higher than those evaluated in our previously completed Phase 2 clinical trials. Phase 1 multiple-ascending dose study is complete, the Company expects that a full analysis of the complete PK data set from the phase 1 study will help determine the choice of any other potential indication(s) for further development of fezagepras. No dose-limiting adverse events or other potential safety signals have been observed in the Phase 1 multiple-ascending dose study to date.

Liminal BioSciences has active business operations in Canada and the United Kingdom.

Forward Looking Statement

This press release contains forward-looking statements about Liminal BioSciences' objectives, strategies and businesses that involve risks and uncertainties. Forward-looking information includes statements concerning, among other things, statements with respect to: the closing of the share purchase agreement for the Ryplazim® business; the sale of the PRV or receipt of proceeds from such sale; the utilization of proceeds from any such transaction; the potential of our product

Press Release for immediate release



candidates and development of R&D programs and the timing of initiation or nature of preclinical and clinical trials.

These statements are "forward-looking" because they are based on our current expectations about the markets we operate in and on various estimates and assumptions. Actual events or results may differ materially from those anticipated in these forward-looking statements if known or unknown risks affect our business, or if our estimates or assumptions turn out to be inaccurate. Among the factors that could cause actual results to differ materially from those described or projected herein include, but are not limited to, risks associated with: the closing of the share purchase agreement; the Company's ability to monetize the PRV, or develop, manufacture, and successfully commercialize product candidates, if ever; the impact of the COVID-19 pandemic on the Company's business operations, clinical development, regulatory activities and financial and other corporate impacts; the availability of funds and resources to pursue R&D projects, manufacturing operations or commercialization activities; the successful and timely completion of clinical trials; the ability of Liminal BioSciences to take advantage of financing opportunities or business opportunities in the pharmaceutical industry; uncertainties associated generally with research and development, clinical trials and related regulatory reviews and approvals; and general changes in economic conditions. You will find a more detailed assessment of these risks, uncertainties and other risks that could cause actual events or results to materially differ from our current expectations in the filings the Company makes with the U.S. Securities and Exchange Commission and Canadian Securities Commissions filings and reports filings and reports, including in the Annual Report on Form 20-F for the year ended December 31, 2020 and future filings and reports by the Company, from time to time. Such risks may be amplified by the COVID-19 pandemic and its potential impact on Liminal BioSciences' business and the global economy. As a result, we cannot quarantee that any forward-looking statement will materialize. Existing and prospective investors are cautioned not to place undue reliance on these forward-looking statements and estimates, which speak only as of the date hereof. We assume no obligation to update any forward-looking statement contained in this Press Release even if new information becomes available, as a result of future events or for any other reason, unless required by applicable securities laws and regulations.

Press Release for immediate release



For further information please contact:

Corporate Contact

Shrinal Inamdar
Manager, Investor Relations and Communications
s.inamdar@liminalbiosciences.com
+1 450.781.0115

Media Contact

Kaitlin Gallagher kgallagher@berrypr.com +1 212.253.8881

Press Release for immediate release

CERTAIN IDENTIFIED INFORMATION MARKED WITH [***] HAS BEEN EXCLUDED FROM THE EXHIBIT BECAUSE IT IS BOTH NOT MATERIAL AND IS THE TYPE THAT THE REGISTRANT TREATS AS PRIVATE OR CONFIDENTIAL.

SHARE PURCHASE AGREEMENT

among

Kedrion S.p.A. as Purchaser

-and-

Liminal BioSciences Inc. as Seller

dated June 22, 2021

TABLE OF CONTENTS

ARTICLE 1	INTERPRETATION	1
------------------	----------------	---

<u>1.1</u>	Definitions. 1
1.2	Other Defined Terms. 9
1.3	Headings. 11
1.4	Extended Meanings. 11
1.5	Ambiguity. 11
1.6	Statutory References. 11
1.7	Accounting Principles. 12
1.8	Currency. 12
1.9	Schedules and Exhibit. 12
1.10	Knowledge. 12
	ARTICLE 2 RUPCHASE AND SALE 42
	ARTICLE 2 PURCHASE AND SALE 12
2.1	Agreement of Purchase and Sale. 12
2.2	PBP Purchase Price. 12
2.3	PBT Purchase Price. 13
2.4	Payment of Purchase Price. 13
2.5	Withholding. 13
	ARTICLE 3 ADJUSTMENTS 13
3.1	Delivery of Estimate Statement. 13
3.2	Delivery of Closing Statement. 13
3.3	Net Indebtedness and Working Capital Adjustment. 14
3.4	Adjustment of Purchase Price for Plasma Inventory 15
	ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF SELLER AS TO ITSELF 15
4.1	Representations and Warranties of Seller as to Itself. 15
Δ	ARTICLE 5 REPRESENTATIONS AND WARRANTIES OF SELLER AS TO THE CORPORATIONS 16
<u>5.1</u>	Representations and Warranties of Seller as to the Corporations. 16
<u>5.2</u>	Survival of Representations. 32
<u>5.3</u>	Non-Ryplazim Assets. 32
<u>5.4</u>	Data Room Documents. 32
	ARTICLE 6 REPRESENTATIONS, WARRANTIES OF PURCHASER 32
6.1	Representations and Warranties of Purchaser. 32
0.1	Nepresentations and Warranties or Furchaser. 32
	ARTICLE 7 PRE-CLOSING COVENANTS OF THE PARTIES 34
<u>7.1</u>	Conduct of Business Prior to Closing. 34
7.2	Access for Due Diligence. 36
7.3	Sale of the PRV and PRV Reorganization 36
7.4	Pre-Closing Reorganization. 37
7.5	Notice of Untrue Representation or Warranty 38

7.6	Confidentiality. 38	
7.7	D&O Indemnification and Insurance.	<u>38</u>
7.8	Notice of Events; Updating Schedules.	38
113422456 (i)		

	Intellectual Property License. 39	
<u>7.10</u>	Intercompany Indebtedness. 39	
<u>7.11</u>	Exclusivity. 39	
7.12	Activities and Costs. 39	
	ARTICLE 8 POST-CLOSING COVENANTS AND AGREEMENTS	40
8.1	Books and Records. 40	
8.2	Further Assurances. 40	
8.3	Regulatory Notices 40	
8.4	Transfer Taxes. 40	
8.5	Consents, Notifications Waiting Periods. 41	
8.6	Commercialization of Ryplazim and Compliance with PRV Legal Requirements	41
8.7	Non-Ryplazim Assets 41	
	ARTICLE 9 CONDITIONS OF CLOSING 41	
9.1	Conditions for the Benefit of Purchaser 41	
9.2	Conditions for the Benefit of Seller. 43	
9.3	Mutual Conditions. 44	
	ARTICLE 10 CLOSING 44	
10.1	Closing. 44	
10.2	Closing Procedures. 44	
	ARTICLE 11 TERMINATION 45	
<u>11.1</u>	Termination Rights. 45	
11.2	Effect of Termination. 45	
	ARTICLE 12 INDEMNIFICATION 46	
12.1	Purchaser Indemnification. 46	
12.2	Seller Indemnification. 47	
12.3	Time Limitations. 47	
12.4	Limitations on Indemnification. 48	
12.5	Notification. 50	
<u>12.6</u>	Direct Claim 50	
12.7	Defense of Third Party Claim. 50	
12.8	Exclusion of Other Remedies 51	
12.9	One Recovery 52	
12.10	Duty to Mitigate 52	
12.11	Adjustment to Purchase Price 52	
	ARTICLE 13 TAX MATTERS 52	
13.1	Tax Returns. 52	
13.2	Tax Cooperation. 53	
13.3	Conduct of Tax Matters. 54	
13.4	Conflicts; Survival. 55	
13.5	Excessive Dividend Elections. 55	
<u>13.6</u>	§338(g)Election. 55	

<u>55</u>

113422456

(ii)

14.1	OFAC Representation. 55	
14.2	Purchaser's Assurances. 56	
14.3	Anti-Money Laundering. 56	
	ARTICLE 15 GENI	ERAL 56
15.1	Fees and Expenses. 56	
<u>15.2</u>	Further Assurances. 56	
<u>15.3</u>	Public Announcements. 57	
15.4	Benefit of the Agreement. 57	
<u>15.5</u>	Entire Agreement. 57	
<u>15.6</u>	Amendments and Waivers. 57	
<u>15.7</u>	Assignment. 57	
<u>15.8</u>	Time of Essence. 57	
<u>15.9</u>	Notices. 58	
<u>15.10</u>	No Third Party Beneficiaries. 58	
<u>15.11</u>	Governing Law and Attornment. 59	
<u>15.12</u>	Non-Merger. 59	
<u>15.13</u>	Counterparts. 59	
<u>15.14</u>	Severability. 59	
<u>15.15</u>	Kedrion Code of Conduct. 59	
15.16	Force Majeure. 59	

Exhibit "A" Illustrative Calculation of Working Capital and Net Indebtedness

Exhibit "B" Plasma Specifications

Exhibit "C" Forms of Certificate of Analysis for the Plasma Inventory

113422456

(iii)

SHARE PURCHASE AGREEMENT

Share Purchase Agreement dated June 22, 2021 among, Kedrion S.p.A. ("Purchaser" or "Kedrion") or a designee thereof as may be provided in accordance with Section 15.7 of this Agreement, and Liminal BioSciences Inc. ("Seller").

WHEREAS Seller is the beneficial and registered owner of one hundred percent (100%) of the issued and outstanding shares of each of Prometic Biotherapeutics Inc. ("PBT") and Prometic Bioproduction Inc. ("PBP", and together with PBT, the "Corporations" or "Ryplazim Targets");

WHEREAS Seller wishes to sell the shares representing all equity interests in the Corporations to Purchaser and Purchaser wishes to purchase such shares from Seller upon and subject to the terms and conditions set out herein; and

WHEREAS Seller granted Purchaser an exclusive option to acquire the shares representing all of the equity interests in the Ryplazim Targets and Kedrion acquired an exclusive option to acquire all such shares (the "**Option**") upon and subject to the terms and conditions set out in that certain option agreement between Purchaser and Seller dated May 21 2021 (the "**Option Agreement**");

NOW THEREFORE, in consideration of the premises and the mutual agreements and covenants hereinafter set forth, Purchaser and Seller hereby agree as follows:

ARTICLE 1 INTERPRETATION

1.1 Definitions.

As used in this Agreement, the following terms have the following meanings:

"Affiliate" means, with respect to any Person, any (i) other Person directly or indirectly controlled by or under common control with such first Person, or (ii) any Person controlled by or under common control with any Person referred to in (i) above. For the purposes of this definition, "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that Person, whether through ownership of voting securities, by contract or otherwise;

"Agreement" means this share purchase agreement, including its Preamble, Recitals, Exhibits and Schedules, as amended, supplemented or restated in writing from time to time, subject to Section 15.6;

"BLA" means the biological license application for Ryplazim plasminogen (human) for the treatment of clinical signs and symptoms associated with congenital plasminogen deficiency;

"Books and Records" means all books, records, files, customer and supplier lists, business reports, plans, projections and all other documentation, correspondence and other information (in whatever medium and wherever situated, including all data and information stored electronically or on computer-related media) relating exclusively to the Business or the Corporations in the possession or under the control of the Corporations or Seller;

"Business" means the business conducted by the Corporations of developing, manufacturing and distributing plasma-derived Ryplazim (plasminogen) for the use stipulated in the BLA;

"Business Day" means any day other than a Saturday, Sunday or other day on which the principal commercial banks in Milan, Italy, Montreal, Quebec or New York, New York are not open for business during normal business hours;

"Cash" means with respect to the Corporations, all cash and cash equivalents determined in accordance with IFRS;

"cGMP" means current Good Manufacturing and General Biologic Products Standard as defined in 21 CFR Parts, 201, 211, 600 and 606.

"Claims" means any claim, demand, complaint, grievance, safety violation, action, cause or right of action, or any proceeding, arbitration, mediation or other dispute resolution procedure, or any orders, writs, injunctions or decrees;

"Closing" means the completion of the Transaction in respect of the First Closing in regard to PBP or the Second Closing in regard to PBT, respectively, pursuant to this Agreement;

"Closing Time" means the time on the applicable Closing Date at which Closing occurs;

"Code" means the Internal Revenue Code of 1986, in each case including the regulations promulgated thereunder, as amended from time to time;

"Constating Documents" means the articles of incorporation, amalgamation or equivalent, as the case may be, and by-laws of the Corporations;

"Contract" means all material contracts of the Corporations, as follows:

- (i) any distribution, sales, supply, purchase, advertising agency, franchise or manufacturer's representative Contract;
- (ii) any ongoing Contract for the purchase of materials, supplies, equipment or services with binding commitments of more than [***] over the term of the Contract;
- (iii) any Contract necessary for the conduct of the Business, that expires, or that may be renewed at the option of any Person other than the Corporations so as to expire, more than one (1) year after the date of this Agreement;
- (iv) any note, loan or credit agreement, mortgage, security agreement, letter of credit, obligation, agreement, instrument or other Contract for or relating to debt and other Indebtedness having an outstanding amount of at least [***], or any guarantee by a Corporation of third-party obligations or any Contract providing for the creation of any Encumbrance (other than Permitted Encumbrances);
- (v) any Contract with binding commitments for capital expenditures in excess of [***] in the aggregate;
- (vi) any Contract pursuant to which a Corporation is a lessor of any machinery, equipment, motor vehicles, office furniture, fixtures or other movable or personal property with binding commitments involving more than [***] over the term of the Contract;
- (vii) any Contract limiting the freedom of a Corporation to engage in any line of business, compete with any other Person, solicit employees or clients;
- (viii) any Contract involving any continuing representation, warranty or indemnification obligation of a Corporation to any other Person, other than in the ordinary course of the Corporation's business;

- (ix) any partnership, joint venture, limited liability company or alliance Contract; or
- (x) any Contract with any Person with whom a Corporation does not deal at arm's length within the meaning of the Tax Act;
- (xi) any Contract that is material to the conduct of the Business as presently conducted.

"Corporate Records" means the corporate records of the Corporations including, (i) all Constating Documents; (ii) all minutes of meetings and resolutions of shareholders and directors (and any committees) for the past five (5) years; and (iii) the share certificate book, securities register, register of transfers and register of directors, as applicable;

"Corporation Transaction Expenses" means (i) the fees and expenses of investment bankers, attorneys, accountants and other consultants or advisors incurred by any of the Corporations in connection with the negotiation and execution of this Agreement and the consummation of the transactions contemplated hereby, and (ii) except as disclosed on the Transaction Expenses Schedule, any severance, change in control, termination, retention, incentive, or other similar amounts or benefits payable or due by a Corporation as a result of the transactions contemplated hereby (including the employer portion of any Taxes relating to such payments), in each case of (i) and (ii), to the extent not paid prior to the applicable Closing;

"Costs and Expenses" means all costs and expenses payable pursuant to Section 7.12;

"COVID-19 Actions" means all commercially reasonable actions taken, planned or planned to be taken in order to minimize the effects of events, occurrences, conditions, circumstances, or developments arising directly or indirectly as a result of the COVID-19 outbreak, or actions taken by any Governmental Authority in response thereto, in each case as determined by the board of directors of the applicable Corporation in order to preserve its Business;

"COVID-19 Returns" means any and all returns, reports, records, calculations, declarations, elections, attestations, notices, forms, designations, filings, and statements filed or required to be filed, or required to be kept on file in respect of any COVID-19 Subsidy;

"COVID-19 Subsidy" means the Canada Emergency Rent Subsidy, the Canada Emergency Wage Subsidy, and any other COVID-19 related direct or indirect wage, rent or other subsidy or loan offered by a federal, provincial, state or local Governmental Authority;

"Current Assets" means the sum of, without duplication, all (i) accounts receivable (excluding the current portion of any current deferred income Tax assets); (ii) unbilled accounts receivable; (iii) inventory (including, for avoidance of doubt, the Plasma Inventory, as may be adjusted pursuant to the terms of this Agreement), (iv) Tax receivables, and (v) prepaid and other expenses, and in each case excluding, for the avoidance of doubt, any such items to the extent they are taken into account in the Option Agreement Expenses or Costs and Expenses. "Current Assets" excludes Cash;

"Current Liabilities" means the sum of, without duplication, all (i) accounts payable and accrued liabilities (including current income Tax liabilities but excluding the PRV Taxes), and (ii) deferred revenue and customer deposits. "Current Liabilities" excludes (i) any deferred Tax liabilities; (ii) any Indebtedness; (iii) all Corporation Transaction Expenses to the extent unpaid prior to the applicable Closing and (iv) any current lease Liability under IFRS16;

"Damages" means any Losses, liabilities, damages or out-of-pocket expenses (including reasonable legal fees and expenses) whether resulting from a Claim that is instituted or asserted by a third party or not;

"Data Room" shall mean the electronic data room established by Seller containing documents and materials relating to the Corporations and their respective Businesses as constituted on the applicable Closing Date;

"Employee Plans" means all the employee or retiree benefit, fringe benefit, supplemental unemployment benefit, bonus, incentive, deferred compensation, retention, profit sharing, termination, change of control, pension, retirement, stock option, stock purchase, stock appreciation, health, welfare, medical, dental, disability, life insurance and similar plans, programs, arrangements or practices relating to the current or former directors, officers or employees of the Corporations maintained, sponsored or funded by the Corporations or Seller for the benefit of the Corporations, whether written or oral, funded or unfunded, insured or self-insured, registered or unregistered;

"Employees" means those individuals employed by the Corporations, whether such Persons are employed on a full-time, part-time or temporary basis, active or inactive as of the Closing Date;

"Encumbrance" means any hypothecation, mortgage, charge, pledge, prior claim, security interest, assignment, lien (statutory or otherwise), *Bank Act* (Canada) security, servitude, easement, conditional sale, resolutory condition, title retention agreement or other encumbrance, arrangement or condition of any nature which, in substance, secures payment or performance of an obligation or material adverse claim of any nature whatsoever and, in the case of stock or a membership interest, any limitation on the right to vote such stock or membership interest:

"Environment" means the environment or natural environment as defined in any Environmental Laws and includes air, surface water, ground water, land surface, soil, subsurface strata, stream sediments, ambient air (including indoor air), plant and animal life and any other living organisms and any other environmental medium or natural resource, any sewer system and the environment in any workplace;

"Environmental Laws" means all applicable Laws relating to public health and safety, pollution or the protection of the Environment, including Laws relating to civil responsibility for acts or omissions with respect to the Environment, and all orders, rules and regulations promulgated by any Governmental Authority pursuant to such Laws;

"Estimated PRV Tax Payments" means the total amount of the estimated income tax payments (if any) made by PBT to applicable U.S. federal and/or state Taxation Authorities.

"Excluded Assets" means the assets of the Corporations listed in Schedule 7.4;

"Excluded Contracts" means the contracts of the Corporations listed in Schedule 7.4;

"FDA" means the United States Food and Drug Administration;

"Financial Statement Date" means December 31, 2020;

"Financial Statements" means the unaudited income statements and balance sheets of the Corporations as at and for the fiscal years ended December 31, 2020 prepared in accordance with IFRS, and on a consistent basis, except for the absence of notes thereto and applicable disclosures required by IFRS;

"First Closing" means the completion of the sale of the PBP Shares and associated transactions pursuant to this Agreement;

"Governmental Authority" means any (i) governmental or public department, central bank, court, minister, governor-in-council, cabinet, commission, tribunal, board, bureau, agency, commissioner

or instrumentality, whether international, multinational, national, federal, provincial, state, county, municipal, local, or other; (ii) subdivision or authority of any of the above; (iii) stock exchange; and (iv) quasi-governmental or private body exercising any regulatory, expropriation or taxation authority under or for the account of any of the above;

"Governmental Authorizations" means any order, permit, approval, notification, waiver, license or similar authorization issued to or required by the Corporations from any Governmental Authority;

"GST/HST" means the goods and services tax or harmonized sales tax levied under the Excise Tax Act (Canada) and any similar value-added or multi-staged Tax imposed by any Governmental Authority;

"Hazardous Materials" means any waste, special waste or other substance that is prohibited, listed, defined, designated or classified as, or otherwise determined to be, dangerous, hazardous, radioactive, explosive or toxic or a pollutant or a contaminant under or pursuant to any Environmental Laws, including any mixture or solution thereof, and specifically including petroleum and all derivatives thereof or synthetic substitutes therefor and PCBs, asbestos or asbestos-containing materials;

"IFRS" means the International Financial Reporting Standards as adopted by the International Accounting Standard Board, and interpretations issued by the IFRS Interpretations Committee as the relevant time, applied on a consistent basis;

"Indebtedness" means, collectively (and without duplication) with respect to the Corporations:

- (i) all indebtedness whether or not contingent, for borrowed money, (including reimbursement obligations, whether or not contingent, with respect to drawn letters of credit) or for the deferred purchase price of property or services (other than current trade liabilities incurred in the ordinary course of business and payable in accordance with customary practices, and excluding operating leases required to be accounted for as capital leases under IFRS) including any related prepayment fees or expenses (whether short term or long term);
- (ii) all obligations evidenced by notes, bonds, debentures or other similar instruments;
- (iii) any declared dividend not paid by the Corporations as at the date hereof;
- (iv) indebtedness for the deferred purchase price of assets, property, goods or services and all obligations under any conditional sale, consignment or other title retention or similar agreements relating to property or assets purchased by the Corporations, including any seller notes and "earn-out" payments, whether or not matured, except for usual and customary commissions paid on the sale of assets in the ordinary course of business;
- (v) all off-balance sheet financing, including synthetic leases and project financing;
- (vi) all liabilities secured by an Encumbrance on any property (other than operating leases required to be accounted for as capital leases under IFRS);
- (vii) all capital expenditure commitments;
- (viii) assurances to a creditor against loss, including letters of credit and bankers' acceptances, in each case to the extent drawn upon or payable and not contingent;

- (ix) all accrued interest, fees, prepayment, penalties or other similar obligations with respect to any of the foregoing; and
- (x) all obligations of the type referred to in clauses (i) through (x) above which are guaranteed directly or indirectly by a Corporation;

provided that Indebtedness shall exclude any amount owing between the Corporations solely to the extent any such amount would only become due and owing after the Second Closing.

"Intellectual Property" means the Intellectual Property Rights owned by, or licensed to or otherwise used or held for use in the Business by each of the Corporations;

"Intellectual Property Rights" means all domestic and foreign intellectual property rights including: (i) patents, applications for patents and reissues, divisions, continuations, renewals, extensions and continuations-in-part of patents or patent applications; (ii) copyrights, copyright registrations and applications for copyright registrations; (iii) designs, design registrations, design registration applications, industrial designs, industrial design registrations, industrial design registration applications, integrated circuit topographies, mask works, mask work registrations and applications for mask work registrations; and (iv) trade names, business names, corporate names, domain names, website names and world wide web addresses, common law trademarks, trade mark registrations, trade mark applications, trade dress and logos, and the goodwill associated with any of the foregoing; (v) all software other than "shrink-wrap" software; (vi) all internet domain names and internet web-sites and the content thereof; and (vii) all Proprietary Rights. As used herein, the term "Proprietary Rights" means all categories of confidential and proprietary information, including, without limitation, trade secrets, know-how, inventions, invention disclosures (whether or not patentable and whether or not reduced to practice), inventor rights, reports, quality records, engineering notebooks, models, processes, procedures, drawings, specifications, designs, ingredient or component lists, formulae, plans, proposals, technical data, copyrightable works, financial, marketing, customer and business data, pricing and cost information, business and marketing plans, and customer and supplier lists and information, and all other confidential and proprietary information;

"Interim Financial Statements" means the unaudited balance sheet and the accompanying unaudited income statements of the Corporations for the five (5) month period ended May 31, 2021 prepared in accordance with IFRS, and on a basis consistent with the Financial Statements, except for the absence of notes thereto and applicable disclosures required by IFRS;

"Interim Period" means the period between the close of business on the date of this Agreement and the applicable Closing;

"Landlord" means the owner of the real property subject to a Lease;

"Laws" or "Law" means any and all applicable domestic and foreign (i) laws, constitutions, treaties, statutes, codes, ordinances, principles of common law or equity, orders, decrees, rules, regulations and by-laws, (ii) judgments, orders, writs, injunctions, decisions, awards and directives of any Governmental Authority and (iii) to the extent that they have the force of law, policies, guidelines, notices and protocols of any Governmental Authority;

"Lease(s)" means the lease(s) disclosed in Schedule 5.1(i);

"Leased Premises" means the land and buildings leased by the Corporations pursuant to the Leases;

"Losses" shall mean, in respect of any obligation of any party hereto to indemnify any Person pursuant to the terms of this Agreement, any and all losses, liabilities, obligations, damages,

amounts paid in settlement, fines, penalties, and reasonable out-of-pocket costs, expenses and charges, including, without limitation, reasonable attorneys' fees and other amounts incurred in proceedings relating to Losses or pursuant to enforcing rights hereunder, provided, however, that Losses shall not include, and no party shall be liable for, exemplary or punitive damages, except to the extent that amounts payable pursuant to a Third Party Claim constitute exemplary or punitive damages;

"Material Adverse Effect" means any event, occurrence, condition, fact or change that is material and adverse to the business, operations, assets, liabilities or financial condition of either or both of the Corporations or affects the ability of the Seller or the Corporations to consummate the transactions contemplated hereby. "Material Adverse Effect" shall not include, alone or in combination, any fact, change, circumstance or event arising in connection with: (i) changes in (A) general economic conditions (whether international, national or local), (B) political conditions (including acts of war, declared or undeclared, armed hostilities, terrorism and cybersecurity attacks), (C) the industries or markets in which the Corporations operate, or (D) Laws or accounting standards, or the principles or interpretations thereof applicable to the Corporation; (ii) natural disasters or the COVID-19 pandemic or other epidemic or pandemic outbreaks including any continuation thereof; (iii) the announcement or existence of this Agreement, or the consummation of the transactions contemplated by this Agreement, or the announcement of the identity of the Purchaser, including the loss or termination, or threatened loss or termination of, or adverse change or threatened adverse change in, any relationship of the Corporations or any of its affiliates with any of their customers, suppliers, landlords, Employees or other business relations as a consequence thereof; (iv) any failure of the Corporations to meet internal forecasts or projections; (v) any matter which has been expressly disclosed in the Schedules; (vi) actions or omissions of the Corporations, Seller or any of their Affiliates taken with the prior consent of, or at the request of, Purchaser or consistent with the terms of this Agreement (including the Schedules and Exhibits); or (vii) any action taken by Purchaser or any of its Affiliates; provided, with respect to (A), (B), (C) and (D), that any such event, occurrence, condition, fact or change does not affect the Corporations, taken as a whole, disproportionately in comparison to other participants in the industry and territory in which the Corporations operate:

"Net Indebtedness" means without duplication, as of the applicable date and time, the Indebtedness, less the total Cash and security deposits (to the extent such security deposits will remain in place after the Closing) of the applicable Corporation, as determined in accordance with IFRS, which net amount may be positive or negative. For the purposes of the calculation of the Net Indebtedness of PBT, Cash shall exclude an amount equal to the difference between the PRV Taxes and the Estimated PRV Tax Payments. An illustrative calculation of each Corporation's Net Indebtedness based on the Interim Financial Statements is attached as Exhibit A;

"Non-Ryplazim Assets" means any other assets not related to the Businesses, which shall include the patents, clinical data and contracts in force pertaining to IVIG and plasma derived proteins other than plasminogen;

"Option Agreement Expenses" means any expenses paid by or on behalf of Purchaser or its Affiliates to or on behalf of Seller, the Corporations or their respective Affiliates, pursuant to and in accordance with the Option Agreement.

"Parties" means Purchaser and Seller and a "Party" means either one of them;

"PBP Shares" means all of the issued and outstanding shares in the capital of PBP;

"PBT Shares" means all of the issued and outstanding shares in the capital of PBT;

"Permitted Encumbrances" means as at any time, (i) non-consensual Encumbrance for Taxes, assessments, charges or levies imposed by a Governmental Authority on property not due and delinquent at such time and not published at such time on any register of Encumbrances

established by a Governmental Authority reserves for which are provided for in the Financial Statements, the Interim Financial Statements or the Books and Records (ii) charges and privileges incidental to current operations and Encumbrances claimed or held by any Governmental Authority that have not at the time been filed or registered against the title to the property or served upon the Corporations and that relate to obligations not due or delinquent and (iii) Encumbrances listed and described in Schedule 1.1;

"Person" means a natural person, partnership, corporation, joint stock company, trust, unincorporated association, joint venture or other entity or Governmental Authority, and pronouns have a similarly extended meaning;

"Plasma Inventory" means the normal source plasma owned by either of the Corporations;

"Plasma Specifications" means [***].

"Post-Closing Tax Period" of a Corporation means the first taxation period of the Corporation that commences on or after the Closing Date and each subsequent taxation period;

"Pre-Closing Tax Period" of a Corporation means any taxation period that is not a Post-Closing Tax Period or a Straddle Period;

"Product" means Ryplazim® plasminogen, human-tvmh, which is indicated for the treatment of patients with plasminogen deficiency type 1 (hypoplasminogenemia).

"Product Registration" means, with respect to the Product, all pharmaceutical registrations, marketing authorizations and applicable pricing and reimbursement approvals granted by any Governmental Authority to permit the manufacture, storage, promotion, sale and marketing of the Product.

"PRV" means that certain priority review voucher reference number 125659 granted by the FDA to PBT in connection with PBT's application for the BLA;

"PRV Taxes" means the amount of Taxes payable by PBT as a result of the sale of the PRV, as determined in accordance with Section 7.3(e);

"Purchased Shares" means all of the issued and outstanding shares in the capital of the Corporations;

"Purchaser Core Representations" means the representations and warranties contained in Section 6.1(a) (Incorporation, Authority, Capacity and Qualification) and 6.1(b) (Enforceability);

"Required Consents" means any approval, consent, or waiver required to be procured from any Governmental Authority or third party in order to provide Purchaser with the full benefit of the transactions contemplated in this Agreement;

"Ryplazim MSA" means that certain Intercompany Master Services Agreement dated January 31, 2017 between PBP and PBT;

"Second Closing" means the completion of the sale of the PBT Shares and associated transactions pursuant to this Agreement;

"Seller Core Representations" means the representations and warranties contained in Section 4.1(a) (Incorporation, Authority, Capacity and Qualification), 4.1(b) (Enforceability), 4.1(c) (Title to Purchased Shares), and Section 5.1(a) (Incorporation, Formation and Qualification);

"Statement of Work" means the statement of work to be mutually agreed and signed between the Purchaser and the Seller:

"Straddle Period" means any taxable period which begins before the applicable Closing and ends after the applicable Closing;

"Target Working Capital" means, in respect of each Corporation, [***];

"**Tax Act**" means the *Income Tax Act* (Canada), R.S.C. 1985 (5th Supp.), including the regulations promulgated thereunder, as amended from time to time;

"**Taxation Authority**" means any Governmental Authority that is entitled to impose Taxes or to administer any applicable Law related to Taxes;

"Taxes" means (i) any and all taxes, duties, fees, excises, premiums, assessments, imposts, levies and other charges or assessments of any kind whatsoever imposed by any Governmental Authority including those levied on, or measured by, or described with respect to, income, gross receipts, profits, gains, windfalls, capital, capital stock, production, recapture, transfer, land transfer, license, gift, occupation, wealth, environment, net worth, indebtedness, surplus, sales, goods and services, harmonized sales, use, value-added, excise, special assessment, stamp, withholding, business, franchising, real or personal property, health, employee health, payroll, workers' compensation, employment or unemployment, severance, social services, social security, education, utility, surtaxes, customs, import or export, and including all license and registration fees and all employment insurance, health insurance and governmental plan premiums or contributions; (ii) all interest, penalties, fines, additions to tax or other additional amounts imposed by any Governmental Authority on or in respect of amounts of the type described in clause (i) above or this clause (ii); and (iii) any liability for the payment of any amounts of the type described in clauses (i) or (ii) as a result of any express or implied obligation to indemnify any other Person or as a result of being a transferee or successor in interest to any other Person;

"Transaction" means the purchase and sale of the PBP Shares or the PBT Shares, as applicable and the other transactions contemplated in this Agreement and the other Transaction Documents;

"Transaction Documents" means this Agreement, the Transition Services Agreement, the Statement of Work pursuant to the Ryplazim MSA and any other agreement or document to be delivered under or pursuant to this Agreement;

"Transition Services Agreement" means the transition services agreements to be mutually agreed and signed between the Corporations, Purchaser and Seller;

"WIP Inventory" means the work-in-progress inventory of the Corporations in accordance with IFRS.

"Working Capital" means, with respect to either Corporation and as of the applicable Closing Time, the Current Assets minus the Current Liabilities in each case of the applicable Corporation, calculated in accordance with IFRS applied on a basis consistent with the Financial Statements or Interim Financial Statements as applicable. An illustrative calculation of each Corporation's Working Capital as at May 31, 2021 based on the Interim Financial Statements is attached as Exhibit A.

1.2 Other Defined Terms.

Term	Section
Amended Pre-Closing Returns	13.1(b)
Anti-Money Laundering Laws	14.3

113422456

Term	Section
Applicable Permits	5.1(g)(iii)
Closing Date	10.1
Closing Statement	3.3(a)
Commitment Payments	7.3(b)
Confidentiality Agreement	7.6
Corporations	Recitals
Direct Claim	12.6(a)
Draft Statement	3.2
Environmental Claims	12.4(b)
Estimate Statement	3.1
Estimated Net Indebtedness	3.1
Estimated Working Capital	3.1
Financial Institution	14.1
First Closing Date	10.1
Improvements	5.1(i)(ii)
Indemnification Notice	12.5
Indemnified Party	12.5
Indemnifying Party	12.5
Independent Accountant	3.3(a)
Italian Code	15.15
Joint Controlled Proceeding	13.3(b)
Kedrion	Preamble
Model	15.15
Objection Notice	3.3(a)
Option	Recitals
Option Agreement	Recitals
OFAC	14.1
Outside Date	11.1(ii)(B)
Patriot Act	14.3
PBP Purchase Price	2.2
PBT Purchase Price	2.3
Pre-Closing Portion	12.1(d)
Pre-Closing Reorganization	7.4
Pre-Closing Returns	13.1(a)
Prepaid Amount	13.3(c)
PRV Sale Date	7.3(a)
Purchase Price	2.3
Purchaser	Preamble
Purchaser Closing Certificate	9.2(a)
Purchaser Indemnified Parties	12.1
Real Property Permits	5.1(i)(v)
Ryplazim Targets	Recitals
Second Closing Date	10.1
Seller Indemnified Party(ies)	12.2

Term	Section
Seller	Preamble
Seller Closing Certificate	9.1
Seller Controlled Proceeding	13.3(b)
Specially Designated Nationals and Blocked Persons	14.1
Specified Tax Position	13.1(a)
Straddle Returns	13.1(c)
Tax Assessment Period	12.3(a)(ii)
Tax Matter	13.3(a)
Tax Returns	5.1(r)(i)
Third Party Claim	12.7(a)

1.3 Headings.

The division of this Agreement into Articles and Sections and the insertion of a table of contents and headings are for convenience of reference only and do not affect the construction or interpretation of this Agreement. The terms "hereof", "hereunder" and similar expressions refer to this Agreement and not to any particular Article, Section or other portion hereof. Unless something in the subject matter or context is inconsistent therewith, references herein to Articles, Sections and Schedules are to Articles and Sections of and Schedules to this Agreement.

1.4 Extended Meanings.

For the purposes of this Agreement:

- (a) words importing the singular number include the plural and vice versa, words importing any gender include all genders;
- (b) the term "including" means "including without limiting the generality of the foregoing", the term "include" means "include, without limiting the generality of the foregoing", and the term "third party" means any Person other than Seller and Purchaser and their respective Affiliates; and
- (c) the phrase "in any capacity whatsoever" means "in any capacity whatsoever, including as an employer, employee, principal, director, officer, agent, mandatary creditor, joint venturer, partner, shareholder or other equity holder, independent contractor, distributor, advisor, consultant, supplier or trustee, or by and through any Person or otherwise".

1.5 Ambiguity.

The rule of construction that, in the event of ambiguity, the contract will be interpreted against the Party responsible for the drafting or preparation of the contract, will not apply to this Agreement or to the interpretation of this Agreement.

1.6 Statutory References.

In this Agreement, unless something in the subject matter or context is inconsistent therewith or unless otherwise herein provided, a reference to any statute is to that statute as now enacted or as the same may from time to time be amended, re-enacted or replaced and includes any regulations made thereunder.

113422456

1.7 Accounting Principles.

All accounting terms not specifically defined in this Agreement shall, unless otherwise provided, be interpreted in accordance with IFRS.

1.8 Currency.

All references to currency herein are to lawful money of the United States of America unless otherwise specifically indicated. Any amounts included denominated in a currency other than American dollars will be calculated on an as converted to American Dollars basis using the Bank of Canada as of the close of business on the day immediately prior to the date on which the Estimate Statement is delivered pursuant to Section 3.1.

1.9 Schedules and Exhibit.

All Schedules and Exhibits attached to this Agreement form an integral part of this Agreement.

1.10 Knowledge.

Where any representation or warranty contained in this Agreement is qualified by reference to the knowledge of the Seller or the Corporations, it refers to the actual knowledge (without further inquiry) of [***], [***] and [***], without personal liability on their part.

ARTICLE 2 PURCHASE AND SALE

2.1 Agreement of Purchase and Sale.

Subject to the terms and conditions of this Agreement, (i) at the First Closing, Seller shall sell, assign and transfer the PBP Shares to Purchaser, and Purchaser shall purchase the PBP Shares from Seller, free and clear of all Encumbrances and (ii) at the Second Closing, Seller shall sell, assign and transfer the PBT Shares to Purchaser, and Purchaser shall purchase the PBT Shares from Seller, free and clear of all Encumbrances.

2.2 PBP Purchase Price.

The purchase price payable by Purchaser to Seller in respect of the PBP Shares (the "**PBP Purchase Price**") shall be an amount equal to:

- (a) \$5,000,000;
- (b) minus an amount equal to the Estimated Net Indebtedness of PBP;
- (c) plus an amount equal to the amount (if any) by which the Estimated Working Capital of PBP is greater than the Target Working Capital of PBP; and
- (d) minus an amount equal to the amount (if any) by which the Target Working Capital of PBP is greater than the Estimated Working Capital of PBP.

The PBP Purchase Price will be subject to final adjustments pursuant to and in accordance with Section 3.3, 3.4 and 7.12.

113422456

2.3 PBT Purchase Price.

The purchase price payable by Purchaser to Seller in respect of the PBT Shares (the "PBT Purchase Price" and, together with PBP Purchase Price, the "Purchase Price") shall be an amount equal to:

- (a) \$5,200,000;
- (b) minus an amount equal to the Estimated Net Indebtedness of PBT;
- (c) plus an amount equal to the amount (if any) by which the Estimated Working Capital of PBT is greater than the Target Working Capital of PBP; and
- (d) minus an amount equal to the amount (if any) by which the Target Working Capital of PBP is greater than the Estimated Working Capital of PBT.

The PBT Purchase Price will be subject to final adjustments pursuant to and in accordance with Section 3.3.

2.4 Payment of Purchase Price.

Subject to the satisfaction or waiver of all conditions of the Purchaser and/or Seller contained in this Agreement, (i) on the First Closing Date, Purchaser shall pay to Seller the PBP Purchase Price, and (ii) on the Second Closing Date, subject to the Closing in respect of the acquisition of the PBP Shares having occurred, Purchaser shall pay to Seller the PBT Purchase Price, in each case by wire transfer of immediately available funds to an account designated in writing by Seller to Purchaser in accordance with the terms of this Agreement (subject to any post-Closing adjustments made in accordance with Article 3 and Article 12).

2.5 Withholding.

Purchaser and Seller shall be entitled to deduct and withhold from any payment made pursuant to, or in connection with the transactions contemplated by this Agreement all Taxes that Purchaser and Seller may be required to deduct and withhold under applicable Law. All such withheld amounts shall be treated as having been paid to the Persons with respect to whom such amounts were deducted and withheld.

ARTICLE 3 ADJUSTMENTS

3.1 Delivery of Estimate Statement.

Unless otherwise agreed among the Parties, three (3) Business Days prior to each respective Closing Date, Seller shall provide Purchaser with a statement (the "Estimate Statement") that sets out a good faith estimate in respect of the applicable Corporation of (i) Working Capital at the Closing Time (the "Estimated Working Capital"), including a balance of Current Assets and Current Liabilities included in the calculation of the Estimated Working Capital, and (ii) the Net Indebtedness at the Closing Time ("Estimated Net Indebtedness"), determined in accordance with IFRS, consistently applied, and the provisions of this Agreement in the form annexed hereto as Exhibit A;

3.2 Delivery of Closing Statement.

Within sixty (60) days after each respective Closing Date, Purchaser will cause the applicable Corporation to provide Seller with a statement (the "**Draft Statement**") that sets out the determination of such Corporation's actual Working Capital and Net Indebtedness of the Closing Date, determined in accordance with this Agreement and in accordance with IFRS, consistently applied. Upon reasonable

113422456

request, (i) Seller shall cooperate fully with Purchaser and its representatives to the extent required to prepare the Draft Statement, and (ii) until the finalization of the Closing Statement, Purchaser shall provide to Seller and its representatives reasonable access to all working papers of Purchaser and the Corporations and their respective accounting and financial books and records relating to their respective Businesses and the appropriate personnel to the extent necessary to verify the accuracy, presentation and other matters relating to the preparation of the Draft Statement.

3.3 Net Indebtedness and Working Capital Adjustment.

- Within thirty (30) Business Days of receipt of the Draft Statement, Seller may notify Purchaser in writing (a) of any objections it may have to the Draft Statement and the Working Capital and the Net Indebtedness calculation of the applicable Corporation provided therein (an "Objection Notice"), which Objection Notice will set forth Seller's calculation of the relevant amounts and a detailed description of the nature and basis for the objections. If an Objection Notice is not timely delivered to Purchaser, the Draft Statement shall become the "Closing Statement" for the purposes hereof and the Net Indebtedness and Working Capital set forth in the Draft Statement will be final, conclusive and binding on the Parties. If an Objection Notice is timely delivered to Purchaser, then Seller and Purchaser will forthwith and in any event within thirty (30) days, negotiate in good faith to resolve the objections set forth in the Objection Notice. In the event that Seller and Purchaser are unable to resolve all such objections within thirty (30) days after Purchaser's receipt of such Objection Notice, Seller and Purchaser will submit such remaining disagreements to Richter, and if Richter is unwilling or unable to act (e.g. due to any conflict of interest with either party), to any such other internationally recognized firm that the Parties may mutually agree upon (the "Independent Accountant"). Purchaser and Seller shall use commercially reasonable efforts to cause the determination of the dispute by the Independent Accountant to be made within thirty (30) days of the date of such submission. The Independent Accountant shall allow each of Purchaser and Seller to submit, in writing, their respective positions regarding the Draft Statement, and Purchaser and Seller shall have the right to submit additional documents, materials and other written information with appropriate copies and notifications sent to the other Party at the same time. The Independent Accountant shall consider such additional documents, materials and other written information. Purchaser and Seller shall jointly instruct the Independent Accountant that: (i) it shall act as an expert in accounting, and not as an arbitrator, to resolve the unresolved disputed items and amounts that were properly included in the notice of objection in accordance with this Agreement; (ii) it shall base their decision solely on the written submissions of Purchaser and Seller, and not on independent review; and (iii) it may not assign a dollar value to any disputed item greater than the highest amount or less than the lowest amount claimed by Purchaser and Seller, as applicable. The Independent Accountant's determination of the dispute will be final and binding upon the Parties. The Independent Accountant's determination of Working Capital and Net Indebtedness will be conclusive and binding on the Parties, absent manifest error, and will become the "Closing Statement" for purposes hereof.
- (b) If the Working Capital set forth in the Closing Statement is greater than the Estimated Working Capital set forth in the Estimate Statement, the amount of such difference will be owed to Seller by Purchaser. In which case, Purchaser shall pay, within twenty (20) days of the final determination of the Closing Statement, such amount to Seller. If the Working Capital set forth in the Closing Statement is less than the Estimated Working Capital set forth in the Estimate Statement, the amount of such difference will be owed to Purchaser by Seller. In which case, Seller shall pay, within twenty (20) days of the final determination of the Closing Statement, such amount to Purchaser.
- (c) If the Net Indebtedness set forth in the Closing Statement is greater than Estimated Net Indebtedness set forth in the Estimate Statement, then the amount of such difference shall be paid by Seller to Purchaser within twenty (20) days. If the Net Indebtedness set forth in

- the Closing Statement is less than the Estimated Net Indebtedness set forth in the Estimate Statement, then the amount of such difference shall be paid by Purchaser to Seller within twenty (20) days.
- (d) Seller, on the one hand, and Purchaser, on the other hand, will each bear the fees and expenses of their respective accountants, auditors and other professional advisors in preparing, reviewing or settling, as the case may be, the Draft Statement. In the case of a dispute and the retention of the Independent Accountant to determine such dispute, the fees and expenses of the Independent Accountant will be borne equally by Seller, on the one hand, and Purchaser, on the other hand.

3.4 Adjustment of Purchase Price for Plasma Inventory

Within one hundred twenty (120) days following the date of the First Closing, Purchaser and Seller or their respective designees shall jointly conduct an inspection and audit of

- the Plasma Inventory, to determine the amounts, types and conformance of the Plasma Inventory with the Plasma Specifications. [***]. Any non-conforming Plasma Inventory shall be delivered to Seller, an Affiliate or third party purchaser or storage provider designated by Seller, or destroyed if required by applicable Law and so requested by Seller, at the sole cost and expense of Seller and in accordance with applicable Law within ten (10) days of the completion of the joint inspection and audit. In the event the Parties fail to agree whether or not any Plasma Inventory conforms to the Plasma Specifications, then the matter will be promptly referred to an independent expert agreed in good faith by the Parties, whose decision shall be binding on the Parties and shall be non-appealable. The cost of referral to the expert will be borne by the Party determined to have been in error as to conformity, or lack thereof, with the Plasma Specifications; and
- (b) the WIP Inventory, to determine the amounts, types and conformance of the WIP Inventory with the statements and approvals in the dossier submitted to the FDA in respect thereof. In the event any WIP Inventory is not compliant with such dossier, such non-conforming WIP Inventory shall not be acquired by Purchaser, and the value of the WIP Inventory shall be reduced the per-lot value for such portion of non-conforming WIP Inventory.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF SELLER AS TO ITSELF

4.1 Representations and Warranties of Seller as to Itself.

Seller represents and warrants to Purchaser as follows and acknowledges and agrees that Purchaser is relying upon such representations and warranties in connection with the purchase by Purchaser of the Purchased Shares:

- (a) **Incorporation, Authority, Capacity and Qualification**. Seller is duly incorporated, organized, in good standing and existing under the Laws of its jurisdiction of incorporation and has the requisite power and authority to enter into and perform its obligations under this Agreement and each of the other Transaction Documents to which it is a party.
- (b) **Enforceability**. This Agreement has been duly executed and delivered by, and constitutes a legal, valid and binding obligation of Seller enforceable against Seller, in accordance with its terms subject only to any limitation under applicable Laws relating to (i) bankruptcy, winding-up, insolvency, arrangement and other Laws of general application affecting the enforcement of creditors' rights and (ii) the discretion that a court may exercise in the granting of extraordinary remedies such as specific performance and injunction.

- (c) **Title to Purchased Shares**. Seller is the sole and unconditional owner of the Purchased Shares set forth beside its name in <u>Schedule 4.1(c)</u>, with a good and valid title thereto, free and clear of all Encumbrances, except for the Permitted Encumbrances. At the Closing Time, Seller will transfer good and valid title to all of the Purchased Shares to Purchaser, free and clear of all Encumbrances.
- (d) Other Agreement to purchase the Purchased Shares. Except for the rights in favor of the Purchaser provided in the Option Agreement, there are no options or other agreements outstanding to purchase any of the Purchased Shares.
- (e) Residence of Seller. Seller is not a non-resident of Canada for the purposes of the Tax Act.
- (f) **Solvency of Seller**. No order has been made, petition presented or resolution passed for the winding up of the Seller or the Corporations or for the appointment of a liquidator or provisional liquidator to the Seller or the Corporations. No administrator has been appointed in relation to the Seller or the Corporations. No notice has been given or filed with the court of an intention to appoint an administrator. No petition or application has been presented or order made for the appointment of an administrator in respect of the Seller or the Corporations. The Seller and Corporations have not proposed or agreed to a general composition, compromise, assignment or arrangement with any of its creditors.

ARTICLE 5 REPRESENTATIONS AND WARRANTIES OF SELLER AS TO THE CORPORATIONS

5.1 Representations and Warranties of Seller as to the Corporations.

Seller represents and warrants to Purchaser the representations and warranties set forth below as of the date hereof and as of the Closing Date, and acknowledges and agrees that Purchaser is relying on such representations and warranties in connection with the Transaction:

- (a) **Incorporation, Formation and Qualification**. Each of the Corporations is incorporated and existing under the Laws of its jurisdiction of incorporation (PBP is a Canadian corporation and PBT is a Delaware corporation) and has the corporate power and authority to own and operate its property, conduct its Business as currently conducted. The Corporations are qualified, licensed or registered to carry on business in the jurisdictions listed in <u>Schedule 5.1(a)</u>. True, correct and complete copies of the Constating Documents of the Corporations are contained in the Data Room.
- (b) **No Conflict**. Except for the Governmental Authorizations disclosed in <u>Schedule 5.1(c)</u> and the Required Consents disclosed in <u>Schedule 5.1(c)</u>, the execution, delivery of, and performance by Seller of the transactions contemplated by this Agreement does not,:
 - (i) conflict with, violate or breach any provision of any Constating Document of a Ryplazim Target;
 - (ii) except for the Required Consents, result in a violation of, conflict with or constitute in any material respect or result in (in each case, with or without notice, lapse of time or both) a material default or material violation or breach under, or result in the acceleration, termination or cancellation of (or entitle any Person or give any Person the right to accelerate, terminate or cancel) any material obligation under, or result in the loss of a material benefit under, or give rise to any obligation of the Ryplazim Targets to make any payment under any contract to which such Ryplazim Target is a party or by which any of its property or assets is bound, or require any material consent, approval or authorization under, any contract to

which a Ryplazim Target is a party or by which any of its property or assets is bound;

- (iii) contravene or violate in any material respect any Law applicable to the Ryplazim Targets or by which it or any of its assets or properties is bound, or any governmental order to which such Ryplazim Target is a party or by which it or any of its assets or properties is bound, except for any such matters or consequences which would not materially adversely affect the ability of the Ryplazim Target to consummate the transactions contemplated by this Agreement;
- (iv) result in the creation or imposition of any Encumbrance, on any of the assets or properties of the Ryplazim Target; or
- (v) to the Seller's knowledge, result in termination or any impairment of, or require any material consent, approval or authorization under any material permit or license held by the Ryplazim Targets.
- (c) Required Authorizations and Consents. Except for the Governmental Authorizations disclosed in Schedule 5.1(c) and the Required Consents disclosed in Schedule 5.1(c), the execution, delivery and performance by Seller of this Agreement and the other instruments contemplated hereby and the consummation of the transactions contemplated hereby and thereby, do not and will not require any material approval, consent, authorization or act of, or the making by the Ryplazim Targets of any declaration, filing or registration with, or notification to any Governmental Authority, except for which the failure to obtain such approval or consent or acts of non-compliance which would not have a material adverse effect.
- (d) Authorized and Issued Capital. Schedule 5.1(d) sets forth the authorized capital of the Corporations and all of the issued and outstanding shares in the capital of the Corporations. All of such shares are outstanding as fully paid and non-assessable, and are registered in the name of Seller as set forth in Schedule 5.1(d). Such shares have been issued in compliance with all applicable Laws. All of the Purchased Shares are duly authorized, validly issued and will be free and clear of all Encumbrances as of the Closing Time and not issued in violation of pre-emptive rights, and constitute all of the issued and outstanding shares of capital stock of the Corporations. No other capital stock, equity interests or other securities of the Corporations, and no securities or other interests directly or indirectly convertible into, or exchangeable or exercisable for, any such stock, equity interests or securities, are issued or outstanding, nor are the Corporations under any contractual obligation to issue such stock, equity interests or securities. All of the shares are held beneficially and of record by Sellers and will be free and clear of all Encumbrances as of the Closing Time. The Corporations have no subsidiaries and do not own, of record or beneficially, control or have the right to acquire any shares of capital stock, securities convertible into capital stock, or any other security or interest in any other Person.

(e) Books and Records. Financial Statements.

- All Books and Records are complete and accurate in all material respects.
- (ii) The Financial Statements are complete and accurate in all material respects and have been prepared in accordance with IFRS consistently followed throughout the periods indicated. The Ryplazim Targets have maintained Books and Records and Financial Statements which, in reasonable detail, accurately and fairly reflect all material transactions entered into by the Ryplazim Targets, or to which a Ryplazim Target is a party. Complete copies of the Books and Records of the Ryplazim Targets thereof have been placed in the Data Room. Seller has uploaded in the Data Room true, complete and correct copies of the unaudited balance sheet and,

- , income statements of each of the Ryplazim Targets as of year-end 2020 and for the five-month period ending May 31, 2021 (the "Balance Sheet Date") (the "Financial Statements"). The Financial Statements present a true and fair view in all material respects of the financial condition and results of operations of the Ryplazim Targets, as of the dates thereof or for the periods referred to therein. Each of the above statements have been prepared in accordance with the Books and Records of the Ryplazim Targets and in accordance with IFRS and, to the extent not in accordance with IFRS, with the normal accounting practices of the Ryplazim Targets, consistent with past practices and as disclosed in Schedule 5.1(f)(i).
- (iii) Each of the Corporations maintains accurate Books and Records and Financial Statements in all material respects reflecting its assets and liabilities and maintain proper and adequate internal accounting controls that provide reasonable assurance that: (i) transactions are executed with management's authorization; (ii) transactions are recorded as necessary to permit preparation of its financial statements and to maintain accountability for its assets; (iii) access to its assets is permitted only in accordance with management's authorization; (iv) the reporting of its assets is compared with existing assets at regular intervals; and (v) accounts, notes and other receivable and inventory (including the Plasma Inventory) are recorded accurately in all material respects, and proper and adequate procedures are implemented to effect the collection thereof on a current and timely basis.
- (iv) Neither of the Corporations has any claims or liabilities, whether primary or secondary, direct or indirect, absolute, accrued, contingent or otherwise, except for such claims or liabilities set forth in the Financial Statements, further executory liabilities arising under any material contract included in the Disclosure Schedules (and other than as a result of breach thereof), accounts payable to trade creditors and accrued expenses incurred subsequent to the Balance Sheet Date in the ordinary course of business consistent with past practice that are not past due and which have not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (v) Schedule 5.1(e)(v) sets forth a true and correct schedule of all Intercompany Indebtedness and, except as set forth in Schedule 5.1(e)(v), there is no Indebtedness between the Seller and its Affiliates, on the one side and the Corporations, on the other side.
- (vi) The Corporations have not received any material comment, complaint, allegation, assertion or claim regarding their accounting or auditing practices, procedures, methodologies or methods or their respective internal accounting controls (or any significant deficiencies or material weaknesses therein).
- (f) **Conduct of Business in the Ordinary Course**. Except as disclosed in <u>Schedule 5.1(f)</u>, since the Financial Statement Date, each of the Corporations has carried on its Business in all material respects, in the ordinary course of normal day to day operations of the business consistent with past practices.
- (g) Compliance with Laws.
 - (i) Each of the Corporations is conducting its business in material compliance with all applicable Laws. At all times in the past, to the knowledge of the Seller the Corporations have materially complied with, and the Corporations are now in material compliance with, and the Business of the Corporations is being conducted, and has been conducted in the past, in material compliance with all applicable Laws and orders of all Governmental Authorities.

- (ii) The Corporations have complied in all material respects with all applicable Law and applicable Permits, except for such noncompliance as has been cured to the satisfaction of each Governmental Authority having jurisdiction thereof.
- (iii) The Corporations (or their designated agents) have filed or caused to be filed with the FDA, as applicable, all material notices, supplements, updates and annual and other reports, including adverse experience reports, required to be filed with respect to the Product.
- (iv) Neither Corporation has received any written notice that any Governmental Authority (including, but not limited to, the FDA) has commenced or threated to initiate any action against such Corporation or with respect to the any of the products of the Corporations. Neither Corporation has received any Notices of Inspectional Observations (Form FDA 483), establishment inspection reports, notices of violation, warning letter, recall letter or other similar documents from the FDA or any other Governmental Authority having jurisdiction.
- (v) The Corporations have all permits, approvals, registrations, licenses, grants, authorizations, certifications, exceptions, orders and consents, issued by Governmental Authorities with respect to their operations as presently conducted in all material respects including all those required for the marketing of Ryplazim in the United States of America (the "Applicable Permits"). Schedule 5.1(g) lists all applicable Permits, and approvals maintained by the Corporations, subject to receipt of any required approvals of Governmental Authorities, and Seller has provided to Purchaser true, correct and complete copies of the Applicable Permits and all related files, reports, and filings and correspondence with Governmental Authorities in possession of the Seller, the Corporations or uploaded to the Data Room.
- (vi) The Corporations are in compliance in all material respects with all applicable provisions of the United States Federal Food, Drug and Cosmetic Act of 1938, as amended.
- (vii) The Corporations are in compliance with the following specific requirements relating to product candidates and activities in Canada and to applicable foreign jurisdictions, including the United States: (A) the Product complies in all material respects with any conditions of approval and the terms of the applications, if any, submitted by or on behalf of the Corporations to the United States Food and Drug Administration; and (B) all adverse events that were required to be reported by Corporations to the FDA have been so reported in a timely manner.
- (viii) All clinical trials of the Product of the Corporations have been, to the knowledge of the Corporations, conducted in accordance with good clinical practices as required by the United States Food and Drug Administration.
- (ix) Except as disclosed in Schedule 5.1(g), all dossiers for the Product Registration are fully upto-date, in eCTD format and with all required variations and renewals having been filed by the relevant due date for filing.
- (x) There are no compliance obligations regarding the Product with respect to Health Canada, and no approval thereof or with respect thereto has been sought, denied, delayed or is pending.
- (h) **Title to the Assets**. Except as disclosed in Schedule 5.1(h), the Corporations own all of the properties and assets that are material to the business of the Corporations and are

reflected as being owned by the Corporations in the Books and Records. The Corporations own such properties and assets free and clear of all Encumbrances, except for Permitted Encumbrances. Except as disclosed in Schedule 5.1(h), no assets that are material to the business of the Corporations are owned by any Person other than the Corporations except for the Leases, the movable or personal properties leased by the Corporations pursuant to the Contracts and the Intellectual Property Rights licensed to the Corporations.

- (i) Leases. The Corporations are not party to, or under any agreement to become a party to, any lease with respect to immovable or real property other than the Leases, complete and accurate copies of which have been uploaded to the Data Room. Schedule 5.1(i) contains a complete and accurate list of all of the Leases, setting out the identity of the Landlords and tenant, the location of the Leased Premises and the term of the Leases (specifying the current expiration date). The Leases are in full force and effect. PBP has valid and subsisting leases in and the right to quiet enjoyment of the premises demised under its lease which constitute the only Leases under which a Ryplazim Target holds a leasehold interest in real estate. PBT is not a party to any real property lease and has terminated its real property lease for premises located in Maryland, placed the premises in the condition and made the termination payment due under the Termination Agreement. Neither Seller nor PBP has received written notice of any default, breach or violation under the Leases or any condition which with the giving of notice or the passage of time would become a default, breach or violation. No Person has so terminated or accelerated performance of the Leases. The Corporations are not in material breach, default or violation of the Leases providing the Landlord thereunder the right to terminate the Lease, except as set forth in Schedule 5.1(i).
 - (i) With respect to the Leases, except as set forth on <u>Schedule 5.1(i)</u>.
 - (A) each Lease is legal, valid, binding, enforceable and in full force and effect and to the knowledge of the Seller, with respect to the Landlord thereunder;
 - (B) subject to any required notice to the Landlord and the delivery to the Landlord by Purchaser and the Corporations of any financial or other information required by the terms of the Lease, the transactions contemplated by this Agreement do not require the consent of the Landlord and will not result in a breach of or default under the Lease;
 - (C) the possession and quiet enjoyment of the premises by the Corporations is not being disturbed in any material respect, and there are no material disputes between the Seller, the Corporations and the Landlord:
 - (D) neither the Seller nor, to the knowledge of the Seller, neither Landlord is in breach or default of any material obligation under Lease, and, to the knowledge of the Seller, no event has occurred or circumstance exists which, with the delivery of notice, the passage of time or both, would constitute such a breach or default of any material obligation, or permit the termination, modification or acceleration of rent under Lease;
 - (E) all rent to be paid and other payments required to be made pursuant to the Leases have been duly paid or made to date, and no rent deferral has been agreed with the Landlord;
 - (F) neither the Seller, the Corporations, nor to the knowledge of Seller either Landlord, owes, nor will owe in the future, any brokerage commissions or finder's fees with respect to the Lease as a result of any commitments or agreements of Seller or the Corporations;

- (G) neither Landlord is an Affiliate of, or otherwise has any economic interest in, the Seller or the Corporations;
- (H) PBP has not subleased, licensed or otherwise granted any Person the right to use or occupy the Leased Premises or any portion thereof; and
- (I) Except as disclosed in Schedule 5.1(i)1(a)(i)(I), the Corporations have not collaterally assigned or granted any security interest or other Encumbrance in the Leases or any interest therein, and have not provided to the Landlord any letter of credit or security deposit and have not made any advance payment of rent in connection with the Leases;
- (ii) To the knowledge of the Seller, all buildings, structures, fixtures, building systems and equipment, and all components thereof included in the Leased Premises (the "Improvements") are in good condition and repair, subject to normal wear and tear.
- (iii) The Corporations have not received written notice of, and to the knowledge of the Seller, there are no condemnation, expropriation or other proceeding in eminent domain, pending or threatened, affecting the Leased Premises or any portion thereof or interest therein.
- (iv) The Corporations have not received any written notice of, nor does Seller have knowledge of, any material violation by the Corporations or the Landlords of any applicable building, zoning, subdivision, health and safety and other land use laws, and, to the knowledge of the Seller, the current use and occupancy of the Leased Premises.
- (v) All certificates of occupancy, permits, licenses, franchises, approvals and authorizations (collectively, the "Real Property Permits") of all Governmental Authorities held by the Corporations relating to the Leased Premises and contained in the Data Room, represent all Real Property Permits required to use or occupy the Leased Premises. To the knowledge of Seller the Corporations have not received any notice from any Governmental Authority threatening a suspension, revocation, modification or cancellation of any material Real Property Permit.
- (vi) The current use of the Leased Premises made by the Corporations and the activities that are carried on therein are authorized pursuant to the applicable Leases.
- (vii) The Corporations do not own, nor have owned any real property since their date of incorporation or formation.
- (j) **Contracts**. Except as set forth in <u>Schedule 5.1(j)</u> or as in Folder 2 (*Plasma Derived Therapeutics*) of the Data Room, neither of the Ryplazim Targets is a party to, or bound by, any of the following Contracts:
 - (i) any contract with respect to the purchase of goods or services involving a binding commitment for a consideration or a cash expenditure in excess of [***];
 - (ii) any note, loan or credit agreement, mortgage, security agreement, letter of credit, obligation, agreement, instrument or other contract for or relating to Indebtedness, or any guarantee by the Ryplazim Targets of third-party obligations or any contract providing for the creation of any encumbrance, except for the Permitted Encumbrances;

- (iii) any collective bargaining agreement or other contract with any labor union or association representing employees of the Ryplazim Targets, Kedrion acknowledging that a union has been created representing a group of approximately 24 Employees responsible for production and maintenance, but no discussions have taken place to negotiate a collective agreement as of the date of the Agreement
- (iv) any joint venture, partnership or limited liability agreement or similar contract.
- (v) any contract which contains any non-competition, or similar obligations or which otherwise prohibits a Ryplazim Target thereof from freely providing services or supplying products to any customer or potential customers;
- (vi) any contract under which any real property is licensed, leased or subleased;
- (vii) any contract relating to any outstanding commitment for capital expenditures outside of the normal course of business or as provided in the 2021 capital expense budget provided to Purchaser;
- (viii) any contract for the acquisition or disposition, directly or indirectly by Ryplazim Targets, of any material assets (excluding the brokering agreement provided in Section 5(f) Option Agreement relating to the sale of the PRV), capital stock or equity of the Ryplazim Targets;
- (ix) any contract currently in effect between a Ryplazim Target and any current or former director, officer, agent or consultant of the Ryplazim Targets, or the spouses or relatives of such persons including the payment of more than [***] per annum.
- (x) any contract pursuant to which a Ryplazim Target is currently obligated to indemnify any other Person, or any other Person is currently obligated to indemnify a Ryplazim Target other than in the ordinary course of business;
- (xi) any contracts for sponsorship agreements;
- (xii) any contracts or other arrangements relating to the Ryplazim Targets which would under their terms require Seller or the Corporations to pay a material penalty, liquidated damages, compensation or any other material payment including by way of indemnity if the Corporations fail to supply products in accordance with the agreed forecast mechanism or other agreed delivery terms (excluding for the avoidance of doubt any amounts payable under general claims for breach of contract); and
- (xiii) any contract pursuant to which (a) any intellectual property rights used by the Ryplazim Targets and material to the Business are licensed to any Person, (b) a Person has licensed any intellectual property to a Ryplazim Target that is used by the Ryplazim Targets.

Neither of the Ryplazim Targets nor, to the knowledge of Seller, any other Person which is a party to any Contract, is in material breach of or default under, and no event has occurred which with or without notice or lapse of time, or both, would become a material breach of or default under, such Contract. To the Seller's knowledge, Each material contract is in full force an effect with respect to the Ryplazim Target party thereto and, to the Seller's knowledge, the other party thereto. Prior to the date of this Agreement, Seller has uploaded in the Data Room a true, complete and correct copy of each Contract.

(k) Intellectual Property.

- (i) Schedule 5.1(k) lists (A) all Intellectual Property owned by a Corporation that is material to the Business of such Corporation as currently conducted, and (B) all Contracts to which a Corporation is a party to or is bound by pursuant to which either Corporation has received a license or other right to use any third party Intellectual Property Rights that are material to the Business as currently conducted, with the exception of commercially available, shrinkwrap, click-wrap and off-the-shelf software licenses in respect of software that is commercially available to the public;
- (ii) Schedule 5.1(k) lists particulars of all registrations and applications for registration of the Intellectual Property owned by each Corporation that is material to the Business of such Corporation as currently conducted;
- (iii) Except as set forth in <u>Schedule 5.1(k)</u>, neither Corporation is a party to or bound by any Contract that limits or impairs its ability to use, sell, transfer or assign or convey, or that otherwise affects, any of the Intellectual Property owned by such Corporation that is material to the Business of the Corporation as currently conducted;
- (iv) to Seller's knowledge, the Ryplazim Targets own, or possess good, valid and marketable title or adequate licenses or other rights to use, all intellectual property used in or necessary to conduct their respective businesses, and perform the material contracts, as currently being conducted;
- (v) to the knowledge of the Ryplazim Targets, no claim is threatened or pending, and no notice has been received that questions the Ryplazim Targets' title to, any intellectual property that is material to the Ryplazim Targets' for the conduct of the Business;
- (vi) to the knowledge of the Seller, the Ryplazim Targets' Intellectual Property used in the Business does not infringe upon, misappropriate, or otherwise violate the rights of any third party's intellectual property;
- (vii) to the knowledge of Seller, no claim is pending or threatened, to the effect that any intellectual property of the Ryplazim Targets that is material to its Business is invalid or unenforceable, and there are no pending, or, to the knowledge of the Seller threatened, interferences, reexaminations, oppositions, cancellation proceedings, or the foreign equivalent thereof by third parties involving any patents of the Ryplazim Targets;
- (viii) to the knowledge of Seller, there is no pending claim of infringement, misappropriation, or other violation, or threatened violation, of any intellectual property and no such claims have been asserted or threatened against by any person against the Ryplazim Targets in the past five (5) years;
- (ix) to the knowledge of Seller, none of the operations of the Ryplazim Targets (including any and all products and services of the Ryplazim Targets) or its existing or previous businesses, as now (or previously) conducted, or performance of any Contract as it has been, or as is currently being, conducted, infringes upon, misappropriates, or otherwise violates any third party intellectual property; and
- (x) the Ryplazim Targets take reasonable measures to protect the confidentiality of their respective proprietary rights.

- (I) **No Undisclosed Liabilities.** The Ryplazim Targets have no material liabilities, obligations Indebtedness of any nature (whether known or unknown, whether accrued, absolute, contingent or otherwise and whether due or to become due) whether or not of a nature required to be reflected in the balance sheet and income statement, except (i) liabilities or indebtedness to the extent reflected or reserved against in the balance sheet, (ii) liabilities incurred since the date of the balance sheet and income statement in the ordinary course of business and consistent with past practice and that are not material in amount.
- (m) Environmental Matters. The Ryplazim Targets are in compliance in all material respects with all applicable Environmental Laws. The Ryplazim Targets have obtained and are in material compliance with all Material Permits required under applicable Environmental Laws and to the knowledge of the Seller, there are no actions pending, threatened, to revoke, suspend, modify or limit any such Material Permit. There are no claims, actions, proceedings, or investigations pending, or to the knowledge of the Seller threatened, against the Ryplazim Targets arising under any Environmental Law except for such claims, actions, proceedings or investigations that have been fully resolved with no future liability or obligation on the part of the Ryplazim Targets. To the knowledge of the Seller, no releases, storage or disposal of regulated substances have occurred in connection with the existing or previous operations or assets of the Ryplazim Targets which are likely to result in imposition of liability for cleanup, personal injury, property damage or natural resource damage and no hazardous materials are located in the ground or the ground water of any property owned, leased or occupied by any of the Ryplazim Targets. Neither Ryplazim Target has not received any written notice from any Person or Governmental Authority that the Ryplazim Targets may be potentially liable under any Environmental Law for response actions or natural resource damages at any location arising out of conditions relating to past or current operations of the Business of the Ryplazim Targets.
- (n) **Governmental Authorizations**. Except as disclosed in <u>Schedule 5.1(n)</u>, the Corporations hold all Governmental Authorizations that are necessary to the conduct of the Business except for acts of noncompliance which would not have a Material Adverse Effect. Each such Governmental Authorization held by the Corporations is valid and in full force and effect. The Corporations are not in breach, default or violation of any such Governmental Authorization held by the Corporations.

(o) Employees.

- (i) The Corporations are, and have been within the past two (2) years, in compliance with all applicable Laws relating to employment and labour matters, employment standards, human rights, labour relations, pay equity, fair employment practices, employment equity, terms and conditions of employment, classification of employees, workers' compensation, occupational safety and health, immigration, employee and data privacy, except for acts of non-compliance which would not have a Material Adverse Effect.
- (ii) The Corporations are not subject to any claim for wrongful dismissal, constructive dismissal or any other claim, arbitration, grievance, complaint or litigation relating to labour, employment, employment standards, occupational health and safety, human right, labour relation, pay equity, fair employment practices, employment equity, terms and conditions of employment, classification of employees, workers' compensation, occupational safety and health, immigration, employee and data privacy discrimination or termination of employment of any Employees or former employees or relating to any failure to hire a candidate for employment.
- (iii) There is no order pursuant to any Law requiring the reinstatement of any Employee or former employee of the Corporations or requiring the taking of any action or the

- refraining from taking any action, in respect of any Employee or former employee of the Corporations.
- (iv) The Data Room includes (1) an accurate list, as of June 14, 2021, of all Employees (without listing names in accordance with applicable Law) setting out (i) their position or title; (ii) their status (i.e. full time, part time, temporary, casual, seasonal, student); (iii) their total annual remuneration, including a breakdown of salary and bonus or other incentive compensation, if any; (iv) other terms and conditions of their employment (other than Employee Plans); (v) whether the employee is a member of a collective bargaining union; (vi) their age; (vii) their total length of employment including prior employment that would affect calculation of years of service for any purpose, including statutory entitlements, contractual entitlements (express or implied), benefit entitlement or pension entitlement; (viii) whether any employee is on any approved or statutory leave of absence and, if so, the reason for such absence and the expected date of return and (2) any employment agreements with such employees and the current rate of compensation (and the portions thereof attributable to salary, bonus, and other compensation respectively) including benefits of each such Employee.
- (v) Except as set forth in <u>Schedule 5.1(o)</u>, the consummation of the transactions contemplated by this Agreement will not, either alone or in combination with another event (i) entitle any Employee to severance pay or any other payment, except as expressly provided in this Agreement, or (ii) accelerate the time of payment or vesting or increase the amount of compensation due to any such Employee. No senior executive or other key employee of the Corporations has provided notice of his or her intention to terminate his or her employment as a result of or following the consummation of the transactions contemplated by this Agreement. No senior executive or other key employee of the Corporations is party to any confidentiality, non-competition, non-solicitation, proprietary rights or other such agreement that would materially restrict the performance of such person's employment duties with the Corporations or the ability of the Corporations to conduct its or their business.
- (vi) Schedule 5.1(o) lists all independent contractors or consultants currently engaged by the Corporations with a commitment for an annual base fee of more than [***] and the services performed by them. The Corporations have not received in the past 12 months any written notice or claim and, to the knowledge of the Seller, any verbal notice, from any Governmental Authority or such independent contractors or consultants, regarding the classification of such independent contractors or consultants.
- (vii) Except as disclosed in <u>Schedule 5.1(o)(vii)</u>, in the past 12 months, there has not occurred or been threatened any strike, slowdown, picketing, work stoppage, concerted refusal to work overtime or other similar labour activity with respect to any Employees. There are no labour disputes currently subject to any grievance procedure, arbitration or litigation and there is no representation petition pending, or to the knowledge of the Seller, threatened with respect to any Employee.
- (viii) The Corporations are in compliance and have been within the past two (2) years with all applicable Laws pertaining to the employment or termination of employment of its employees, except for any failure to so comply that, individually and in the aggregate, could not result in any material liability or obligation on the part of the Corporations or Purchaser or any of their respective Affiliates.
- (ix) All Employees of PBP are hired for an indefinite term and can, subject to any termination clauses in their contractual agreements, have their employment

terminated with reasonable notice or pay in lieu, as may be required under applicable Law. All Employees of PBT are "at will" employees or can have their employment terminated with sufficient notice as may be required under applicable Law.

- (x) The Corporations are in material compliance and have been within the past two (2) years with all provisions of the applicable occupational health and safety Laws and there are no outstanding claims, charges or orders thereunder.
- (xi) The Corporations are in material compliance and have been within the past two (2) years with applicable workers compensation Laws and there are no outstanding assessments, levies or penalties thereunder.
- (xii) The Corporations have materially followed and fully complied with all applicable federal, provincial and local laws, rules, notices, ordinances, regulations, orders and relating to protecting workers, customers and other visitors to the Leased Premises from spreading or contracting COVID-19. The Corporations are not subject to any claim arbitration, grievance, complaint or litigation relating to the failure to provide a safe working environment or accommodation in relation to COVID-19.
- (xiii) All material payments due from the Corporations on account of wages (including but not limited to vacation and overtime pay) or other compensation, and employee health and welfare insurance, workers' compensation and other benefits, have been paid or properly accrued as a liability in the Books and Records of the Corporation.

(p) Employee Plans

- (i) <u>Schedule 5.1(p)</u> and as in Folder 2 (*Plasma Derived Therapeutics*) of the Data Room lists all Employee Plans.
- (ii) The Corporations have made all contributions and paid all premiums in respect of each Employee Plan in a timely fashion in accordance with the terms of each Employee Plan and applicable Laws except where failure to do so would not have a Material Adverse Effect.
- (iii) Except as disclosed in <u>Schedule 5.1(p)</u>, none of the Employee Plans provide for retiree benefits or for benefits to retired employees or to the beneficiaries or dependents of retired employees.
- (iv) Each such Employee Plan has been established, registered, and administered in material compliance with all Laws.
- (v) Seller has delivered or made available to Purchaser copies of all the Employee Plans, together with all related documentation, including funding and management agreements, investment policy, summary plan descriptions, if any.
- (q) [***]
- (r) **Tax Matters**.

Except as set forth in Schedule 5.1(r) and in respect of the PRV Taxes:

- (i) **Filing and Payment**. The Corporations have prepared and duly and timely filed all federal, provincial, state, municipal, local and foreign returns, information statements, forms, elections, designations, slips, notices, schedules, reports and any other filings related to Taxes ("**Tax Returns**") required to be filed by them on or before Closing, and have timely paid all material Taxes which are due and payable within the time required by applicable Laws.
- (ii) **Status of Assessments**. All Tax Returns of the Corporations have been filed for periods ending on or before December 31, 2019, and there are no outstanding waivers of any limitation periods or agreements providing for an extension of time for the filing of any Tax Return, the payment of any Tax or the assessment or reassessment of Taxes.
- (iii) Withholdings and Collections. The Corporations have withheld and collected the amount of all Taxes required to be withheld and collected and have timely paid such amounts when due, in the form required under applicable Laws. Each of the Corporations has duly filed or caused to be filed on a timely basis, subject to applicable extensions, all Tax Returns and all such Tax Returns are complete and correct in all material respects and such Tax Returns reflect accurately all liability for Taxes of each of the Corporations for the periods covered thereby.
- (iv) Each of the Corporations has duly and timely paid all Taxes, including all instalments on account of Taxes, due and payable by it and has paid all assessments and reassessments it has received in respect of Taxes. Each of the Corporations has made full and adequate provision in the Books and Records for all Taxes which are not yet due and payable but which relate to a Pre-Closing Tax Period, and neither of the Corporations has received any refund of Taxes to which it is not entitled.
- (v) No audit, investigation or other Claim by any Taxation Authority is currently ongoing, pending or threatened in writing with respect to any Taxes due from or with respect to the Corporations and there are no matters under discussion, audit or appeal with any Taxation Authority. There are no outstanding objections to any assessment or reassessment of Taxes for which either of the Corporations may be liable. Any Taxes assessed or reassessed have been paid and settled.
- (vi) Except as set forth in Section 5.1(a)(ii), neither of the Corporations has executed any waiver of any statute of limitations on the assessment or collection of Taxes with respect to the Corporations, nor requested, offered to enter into or entered into any agreement or other arrangement, or executed any agreement or waiver now in effect extending the period of time within which: (i) to assess or collect any Taxes; (ii) to file any Tax Return; or (iii) it is required to remit any Taxes or amounts on account of Taxes.
- (vii) Each of the Corporations has duly and timely withheld from each payment made to any Person, including any of its present or former employees, officers and directors, and to all Persons who are non-residents for the purposes of the applicable Laws, all amounts required by Law and has timely remitted such withheld amounts within the prescribed periods to the appropriate Taxation Authority.
- (viii) Each of the Corporations has duly and timely collected all amounts on account of any sales or transfer Taxes, including, with respect to PBP, GST/HST and provincial or territorial sales Taxes, required by Laws to be collected by it and has duly and timely remitted to the appropriate Taxation Authority any such amounts as and when required by Laws. All tax credits claimed by the Corporations

- pursuant to the Excise Tax Act (Canada) or otherwise have been properly and correctly calculated and documented in accordance with Laws and neither of the Corporations has claimed any credit to which they would not have been entitled.
- (ix) Neither of the Corporations has ever been required to file any Tax Return with, and has never been liable to pay or remit any Taxes to, any Taxation Authority outside of (i), in the case of PBP, Canada, and (ii) in the case of PBT, the United States. No Claim has ever been made by a Taxation Authority in a jurisdiction where a Corporation does not file Tax Returns that the Corporation is or may be subject to taxation by that jurisdiction.
- (x) Each transaction between PBP, on the one hand, and any non-resident Person with whom PBP was not dealing at arm's length (within the meaning of the Tax Act), on the other hand, was priced in a manner so as to not result in an adjustment under section 247 of the Tax Act, and PBP has made or obtained records or documents in respect of the transaction that satisfy the requirements of paragraphs 247(4) (a) to (c) of the Tax Act.
- (xi) Neither of the Corporations has claimed any reserve or deduction for Tax purposes if, as a result of such claim, any amount could be included in its or the Purchaser's income for a Post-Closing Tax Period, except to the extent that the Corporation has claimed a similar reserve or applied a similar accounting policy that is taken into account in determining the applicable Purchase Price in the Closing Statement.
- (xii) There are no circumstances which exist and would result in, or which have existed and resulted in, any of section 78 or sections 80 to and including section 80.04 of the Tax Act applying to PBP.
- (xiii) PBP is registered for purposes of the GST/HST and any applicable Laws relating to provincial or foreign sales Taxes in respect of which it is required to be registered.
- (xiv) There are no Encumbrances for Taxes (other than Permitted Encumbrances) upon any assets of the Corporations.
- (xv) Neither of the Corporations is subject to any joint venture, partnership or other arrangement or contract that is treated as a partnership for income Tax purposes in any jurisdiction.
- (xvi) Neither of the Corporations is party to any Tax sharing agreement, and each of the Corporations has no liability for the Taxes of any other Person as a transferee or successor, by Contract, or otherwise.
- (xvii) PBP has complied with subsection 89(14) of the Tax Act in respect of any dividends designated, declared or reported as eligible dividends and does not have any excessive eligible dividend designations.
- (xviii) Each of the Corporations has (i) duly and timely completed and filed all COVID-19 Returns filed or required to be filed by it, and all such returns are complete, correct and accurate in all material respects; (ii) not received any COVID-19 Subsidy amounts to which it was not entitled; and (iii) not deferred any payroll tax obligations as permitted under applicable COVID-19 related measures enacted, promulgated or offered as an administrative relief by a Governmental Authority.

- (xix) PBP, since its incorporation, has only carried on the business that is the development and manufacturing of plasma-derived medicinal products and has carried on no other business that is unrelated thereto.
- (xx) PBT (i) has not changed its annual accounting period within the most recent 48-month period ending June 30, 2021; (ii) has no interest in any pass-through entity or a CFC, in each case as those terms are defined in Rev. Proc. 2006-45; (iii) is not a shareholder of a foreign sales corporation or interest charge domestic international sales corporation; (iv) is not a foreign sales corporation or interest charge domestic international sales corporation; (v) is not and has never been an S corporation as defined in Internal Revenue Code Section 1361; (vi) is not a personal service corporation as defined in Internal Revenue Code Section 441(i); (vii) is not a controlled foreign corporation as defined in Internal Revenue Code Section 957; (viii) is not a tax-exempt organization; (ix) has no election in effect under Internal Revenue Code Section 936; (x) is not a cooperative association within the meaning of Internal Revenue Code Section 1381(a); (xi) does not have a required taxable year; and (xii) has never been a member of a consolidated group.
- (xxi) **Tax Claims.** There are no claims, actions, proceedings, suits, investigations or audits pending or, to the knowledge of the Seller, threatened against any of the Corporations in respect of Taxes.
- (s) **No Brokers, etc.** Except as provided in Section 5(f) of the Option Agreement with respect to the sale of the PRV, no broker, finder, agent or similar intermediary has acted on behalf of Seller or the Corporations in connection with this Agreement or the transactions contemplated hereby other than the sale of the PRV, and there are no brokerage commissions, finders' fees or similar fees or commissions payable by Seller or the Corporations in connection with this Agreement or the transactions contemplated hereby except for the sale of the PRV.
- (t) Accounts Receivable. Except as set forth in Schedule 5.1(t) except to the extent collected, the Ryplazim Targets' accounts and notes receivable are bona fide receivables arising from sales actually made to or services actually performed by the Ryplazim Targets for third parties, are reflected on the books and records of the Ryplazim Targets, arose in the ordinary course of business and the goods and services involved have been, in all material respects, sold, delivered and performed to the account obligors and no further material amounts of goods or services are required to be provided or rendered, as applicable, in order to complete the sales and fully render the services and entitle a Ryplazim Target to collect such accounts receivable in full and the Ryplazim Targets has not received any notice of any material contest, claim, or right of set-off under any such accounts or notes. Seller has placed in the Data Room an aging schedule of the account receivable and its bad debt reserve, which Seller believes is adequate to cover any reasonably anticipated collection losses.
- (u) Licenses and Permits. The Ryplazim Targets own or hold all permits, licenses approvals and authorizations necessary for the conduct of their respective businesses in all material respects as currently conducted (collectively, "Material Permits"). The Material Permits are valid, and to the knowledge of Seller, no Governmental Authority intends to modify, cancel, terminate or not renew any of the Material Permits, whether as a result of the transactions contemplated hereby or otherwise. Each of the Material Permits is set forth on Schedule 5.1(u).
- (v) **Insurance**. The Data Room sets forth each insurance policy subscribed by the Corporations, but excluding global insurance policies subscribed by the Seller (including global policies providing property, casualty, liability and workers compensation coverage),

at any time within the past two (2) years. With respect to each insurance policy identified in the Data Room as currently maintained by the Corporations: (A) the policy is legal, valid, binding, enforceable, and in full force and effect; (B) neither the Corporations, nor to the knowledge of the Seller, any other party to the policy is in breach or default (including with respect to the payment of premiums or the giving of notices), and, to the knowledge of the Seller, no event has occurred which, with notice or the lapse of time, would constitute such a breach or default, or permit termination, modification, or acceleration, under the policy; and (C) no party to the policy has repudiated any provision thereof. There are no insurance claims pending.

- (w) **Warranties.** Neither Corporation has undertaken or assumed any material warranties or guarantees to its customers with respect to its products and services, other than as set forth in the Contracts, and has not received any notice of breach of warranty or guarantee.
- (x) Inventory. (a) All of the inventories (including the Plasma Inventory) reflected in the Financial Statements of the Ryplazim Targets consisted at the dates thereof of a quality and quantity usable and, with respect to finished goods, saleable or compassionate use, in the ordinary course of the Business and none of the inventories (including the Plasma Inventory) was obsolete, damaged or defective, except for slow moving, damaged or defective items that have been written off or written down to net realizable value in accordance with the Ryplazim Targets' internal policies as disclosed to Kedrion, and (b) the values at which the inventories (including the Plasma Inventory) is carried on the Financial Statements reflect the inventory valuation policy of stating inventories at cost or market, whichever is lower, on a first-in, first-out basis.
- (y) **Certain Business Practices**. Neither Seller nor any representative of the Ryplazim Targets has directly or indirectly (a) made any unlawful contribution, gift, bribe, rebate, payoff, influence payment, kickback, or other payment to any Person, private or public, regardless of what form, whether in money, property, or services (i) to obtain favorable treatment for the Ryplazim Targets or to secure any contracts, (ii) to pay for favorable treatment for business of the Ryplazim Targets or to secure any contracts, (iii) to obtain special concessions for the Ryplazim Targets or for such special concessions already obtained, or (iv) on behalf or for the benefit of the Ryplazim Targets and otherwise in violation of any legal requirement, or (b) established or maintained any fund or asset of the Ryplazim Targets that has not been recorded in the books or records of the Ryplazim Targets.
- (z) **Competition Act.** For the purposes of Part IX of the *Competition Act* (Canada), each of (a) the aggregate value of the assets in Canada, determined as of such time and in such manner as may be prescribed, and (b) the sales in or from Canada, determined for such annual period and in such manner as may be prescribed, of the Corporations in each case does not exceed the amount triggering application of Part IX of the *Competition Act* (Canada).
- (aa) **Investment Canada.** Each of the Corporations is not engaged in a "cultural business activity" as defined in the Investment Canada Act.
- (bb) **Private Issuer.** Each of the Corporations is a "private issuer" within the meaning of National Instrument 45-106-*Prospectus Exemptions*.
- (cc) **No Option on the Purchased Shares.** Except for the option granted in favor of the Purchaser pursuant to the Option Agreement, there are no outstanding:
 - (i) agreements, arrangements, understandings, warrants, contracts, options, puts, calls, rights, options, subscriptions or other commitments of any kind to which the Ryplazim Targets or Seller are a party or by which any of them or any of their

respective properties or assets are bound, relating to the sale, purchase, issuance or voting of any shares of capital stock of, or other membership or ownership interests in, or other securities of, the Ryplazim Targets, or

(ii) securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any shares of capital stock of, or other membership or ownership interests in the Ryplazim Targets.

(dd) BLA and PRV.

- (i) With respect to the BLA and the PRV, the application of the PRV was submitted on September 4, 2020 and true and complete copies of all notices submissions and material communications with the FDA in connection thereto are contained in the Data Room.
- (ii) The studies, tests and preclinical and clinical trials conducted by or on behalf of the Corporations in relation to the Product were conducted in accordance in all material respects with experimental protocols, procedures and controls pursuant to accepted professional scientific standards and all applicable Laws; the descriptions of the results of such studies, tests and trials contained in the Data Room fairly present in all material respects the data derived from such studies, tests and trials; except to the extent disclosed in the Data Room, the Corporations are not aware of any studies, tests or trials the results of which the Corporations believe reasonably call into question the study, test, or trial results described or referred to in the Data Room when viewed in the context in which such results are described and the clinical state of development; and the Corporations have not received any notices or correspondence from any Governmental Authority requiring the termination, suspension or material modification of any current or active studies, tests or preclinical or clinical trials conducted by or on behalf of the Corporations.
- (iii) Except as disclosed in Schedule 5.1(dd)(iii) or in the Data Room, there are no studies, tests, preclinical or clinical trials, compassionate use programs or other similar activities conducted, pending or in progress by or on behalf of the Corporations, or which impose financial or regulatory obligations on the part of the Corporations.
- (ee) **Guarantees and Other Third Party Assurances.** Except as disclosed in Schedule 5.1(ee), in relation to the Corporations there are no guarantees, indemnities or other sureties issued to a third party by PBP or PBT, as applicable, in respect of any obligation of a member of the Seller or its Affiliate; and/or (as the context may require).

5.2 Survival of Representations.

The representations and warranties set forth in Section 5.1 shall survive Closing but any claim based thereon must be notified to the Seller within the periods set forth in <u>Section 12.3</u>.

5.3 Non-Ryplazim Assets.

Notwithstanding anything else to the contrary, the Parties agree that the representations and warranties contained herein are limited to the assets owned by the Corporations other than the Non-Ryplazim Assets, which shall be transferred without any representation or warranty. For the avoidance of

113422456

any doubt, Purchaser agrees that it shall have no action or proceeding based the foregoing representations and warranties in relation to the Non-Ryplazim Assets.

5.4 Data Room Documents.

Each of the documents, instruments, reports, certificates and other materials contained in the Data Room is a true, correct and complete copy of the original thereof (except with respect to any redacting), and the redacted portions of such materials do not contain any provisions which in the aggregate would materially and adversely affect the disclosed portion of such materials or the Business.

ARTICLE 6 REPRESENTATIONS, WARRANTIES OF PURCHASER

6.1 Representations and Warranties of Purchaser.

Purchaser represents and warrants to Seller the representations and warranties set forth below as of the date hereof and as of the Closing Date, and acknowledges and agrees that Seller are relying on such representations and warranties in connection with the Transaction:

- (a) **Incorporation, Formation and Qualification**. Purchaser is a corporation duly incorporated, organized and existing under the Laws of its jurisdiction of incorporation and has the corporate power and authority to own and operate its property and enter into and perform its obligations under this Agreement and each of the other Transaction Documents to which it is a party.
- (b) **Authorization**. The execution and delivery of and performance by Purchaser of this Agreement and each of the other Transaction Documents to which it is a party, and the consummation of the transactions contemplated thereby have been duly authorized by all necessary action required under the organizational documents of Purchaser as required.
- (c) **Enforceability**. Each of this Agreement and each other Transaction Document to which Purchaser is a party has been duly executed and delivered by, and constitutes a legal, valid and binding obligation of, Purchaser, enforceable against Purchaser in accordance with its terms subject only to any limitation under applicable Laws relating to (i) bankruptcy, winding-up, insolvency, arrangement and other similar Laws of general application affecting the enforcement of creditors' rights and (i) the discretion that a court may exercise in interpreting enforceability of a provision or in the granting of extraordinary remedies such as specific performance and injunction.
- (d) No Conflict. The execution, delivery and performance by Purchaser of this Agreement and the other Transaction Documents to which it is a party, do not and will not result in a breach, default or violation of, or conflict with, or allow any other Person to exercise any rights under, any term, condition or provision of its Constating Documents, any Contract or any resolutions of its board of directors or shareholders, and will not result in a material breach or violation of any Law.
- (e) Required Authorizations and Consents. Other than the notification of investment under the Investment Canada Act, there is no requirement of Purchaser to make any filing with, give any notice to, or obtain as a condition to the lawful completion of the transactions contemplated by this Agreement or to obtain any Required Consents under any Contract to which it is a party except to the extent that the failure to obtain any such consents would not prohibit the lawful completion of the transactions contemplated by this Agreement.
- (f) **Execution and Binding Obligation.** This Agreement has been duly executed and delivered by Purchaser and constitutes a legal, valid and binding obligation of Purchaser,

enforceable against it in accordance with its terms subject to any limitation under applicable Laws relating to (i) bankruptcy, winding-up, insolvency, arrangement and other Laws of general application affecting the enforcement of creditors' rights and (ii) the discretion that a court may exercise in the granting of extraordinary remedies such as specific performance and injunction.

- (g) **Litigation.** There are no (i) actions, claims, appeals, grievances, suits, or proceedings, (ii) arbitration or alternative dispute resolution processes in progress; or (iii) administrative proceedings; in each case, pending, or to Purchaser's knowledge, threatened against Purchaser, which prohibits, restricts or seeks to enjoin the transactions contemplated by this Agreement.
- (h) Due Diligence by Purchaser. Purchaser acknowledges that it has conducted an independent investigation of the business, operations, assets, liabilities and financial condition of the Corporations. In making the determination to proceed with the transactions contemplated by this Agreement, Purchaser has relied solely on the results of its own independent investigation and the representations and warranties in Article 5 (including the related portions of the Schedules) and set forth elsewhere in this Agreement. Purchaser hereby recognizes and agrees that such representations and warranties and Seller's performance of its obligations hereunder constitute the complete extent of what it considered to be determinant, decisive and material in the decision to proceed with the transactions contemplated by this Agreement.
- (i) **Available Funds.** Purchaser has, or has access to, and will have at Closing funds on hand necessary to pay each Purchase Price and otherwise fulfil its obligations under this Agreement.
- (j) **No Broker.** Kedrion has not taken any action that would cause the Seller to become liable to any claim or demand for a brokerage commission, finder's fee or other similar payment in respect of the transactions provided for in this Agreement.
- (k) **Solvency of Purchaser**. No order has been made, petition presented or resolution passed for the winding up of Kedrion or for the appointment of a liquidator or provisional liquidator to Kedrion. No administrator has been appointed in relation to Kedrion. No notice has been given or filed with the court of an intention to appoint an administrator. No petition or application has been presented or order made for the appointment of an administrator in respect of Kedrion. Kedrion has not proposed or agreed to a general composition, compromise, assignment or arrangement with any of its creditors.

ARTICLE 7 PRE-CLOSING COVENANTS OF THE PARTIES

7.1 Conduct of Business Prior to Closing.

(i) During the Interim Period, other than any COVID-19 Actions and except as expressly provided in this Agreement or with the prior written consent of Purchaser, which consent shall not be unreasonably withheld, delayed or conditioned, Seller shall cause the Corporations to conduct their respective businesses in the ordinary course consistent with past practice.

113422456

- (ii) Without limiting the generality of Section 7.1(i), until the respective Closing Date, Seller shall cause the Corporations to:
 - (A) use commercially reasonable efforts to preserve intact the current business organization of the Corporations, maintain good relations with, and the goodwill of, suppliers, creditors, distributors and all other Persons having business relationships with the Corporations and preserve the value of their businesses;
 - (B) use commercially reasonable efforts to not cause or permit to exist a breach of any representations and warranties of Seller contained in this Agreement and to conduct the Corporations' businesses in such a manner that on the Closing Date such representations and warranties will be true, correct and complete as if they were made on and as of such date.
 - (C) comply in all material respects with the Corporations' obligations under the Leases and the Contracts and comply in all material respects with all applicable Law;
 - (D) maintain in full force and effect the FDA license, the Governmental Authorizations, trade accreditations and all registrations of all Intellectual Property, to the extent applicable;
 - (E) maintain all material structures, equipment and other tangible personal property in their present state of repair, order and condition, except for depletion and ordinary wear and tear; and
 - (F) maintain in full force and effect and comply in all material respects with the Corporations' obligations under the insurance policies and in effect on the date of this Agreement until the Closing Date, or the earlier termination of this Agreement.
- (iii) During the Interim Period, Seller shall cause each of the Corporations not to take any of the following actions, without the prior written consent of the Purchaser, not to be unreasonably withheld, delayed or conditioned, except in connection with the Pre-Closing Reorganization or the sale of the PRV or as otherwise expressly provided in this Agreement:
 - (A) amend its charter or by-laws or similar organizational documents;
 - (B) (a) split or reclassify any shares of capital stock; (b) issue, pledge, transfer, encumber or sell any additional shares of capital stock of any class or series, or any securities convertible into or exchangeable for, or options or rights to acquire, shares of capital stock of any class or series; or (c) repurchase, redeem or otherwise acquire any shares of its capital stock;
 - (C) enter into, amend or terminate or waive any material right under any material Contract, except for (1) material Contracts entered into, amended or terminated in the ordinary course of business consistent with past practice, (2) a brokering fee agreement with an investment bank or financial institution with respect to the sale of the PRV as provided in Section 5(f) of the Option Agreement and related sale agreement with the buyer of the PRV, *provided* that such sale agreement may not impose any unreasonable cost, expense, obligation or other liability on the Corporations or the Business that is not consistent with market

practice for deals of this nature or as required as a result of the Option Agreement or this Agreement, (3) the Transition Services Agreement or (4) the Statement of Work and related amendment to Ryplazim MSA;

(D) acquire any business, material assets or Person, whether by merger, consolidation, purchase of assets or equity interests, in a single transaction or a series of related

113422456

transactions or enter into any new line of business or materially change or terminate any business practice;

- (E) make any capital expenditure outside of the normal course of business or except as authorized by the 2021 budget plan or forecasted capex of the Ryplazim Targets previously delivered to Kedrion and uploaded to the Data Room;
- (F) (a) sell, lease, license or otherwise dispose of, or incur any Encumbrances on, any material assets or material Intellectual Property (and excluding any Excluded Assets or Non-Ryplazim Products); (b) engage in any liquidation, dissolution, restructuring or other reorganization or any merger, recapitalization, business combination or similar material transaction; or (c) or dispose of or disclose to any Person, other than representatives of Kedrion, any material Intellectual Property; other than in the ordinary course of business.
- (G) (a) incur any material indebtedness other than in favor of an Affiliate and which shall be satisfied in full prior to the Closing or (b) make any loans or advances to, or capital contributions or investments in, any person other than Affiliates except loans or advances to customers in the form of trade credit or deferred purchase price arrangements in the ordinary course of business;
- (H) (a) adopt or amend in any material respect, or accelerate the payment or vesting of benefits or amounts payable under, any Employee Plan; or (b) increase or adjust the compensation or fringe benefits or terms of employment of, or enter into any new employment, bonus or incentive arrangement with, any director, officer or employee of the Ryplazim Targets other than normal course of business;
- (I) implement or effect any, new early retirement program, severance program or other program with respect to the employees of the Ryplazim Targets other than in the normal course of business;
- (J) (a) settle or compromise any claim or liability with respect to Taxes, make, change or revoke any election with respect to Taxes, agree to any adjustment of any Tax attribute, surrender any right to claim a refund, reduction or other recovery for any Taxes, consent to any extension or waiver of the limitation period applicable to any Tax claim or Tax Return, amend any Tax Return or enter into any closing agreement or other agreement (including any agreement with applicable Governmental Authority) relating to Taxes, (b) fail to timely pay all Taxes when due or timely file all required Tax Returns, (c) adopt or change any tax accounting principles, methods or practices with respect to Taxes; or (d) adopt or change any method of accounting or accounting practices in any material respect except as required by IFRS;
- (K) settle or compromise any material legal proceeding or material investigation pertaining to the Ryplazim Targets or their respective Business or enter into any consent decree, injunction or similar restraint or form of equitable relief in settlement of any proceeding;
- (L) other than in the ordinary course of business consistent with past practice, fail to manage the payment of accounts, collection of accounts receivable and loan facility borrowing and repayment;
- (M) with respect to any Employees, (a) increase the rate or terms of compensation payable or to become payable to any of them, except for customary annual salary increases for Employees consistent with past practice not to exceed three percent (3%) in the aggregate or as required by Law, an employment

Contract or Employee Plan, or	agree to pay	any pension,	retirement	allowance of	or other	employee	benefit
not required or permitted by La	aw						

or any employment Contract or any Employee Plan or (b) amend or enter into any new employment or severance Contract;

- (N) make any promise or commitments to with any Employee as to the continued employment of such Employee for a specific duration after the Closing Date;
- (O) make any material alteration in the manner of keeping the books, accounts or records of the Corporations, or in the accounting practices therein reflected, except as required by applicable Law;
- (P) knowingly take action that would cause any representation or warranty herein to be untrue or incorrect as of the date when made, or as of the Closing Date or which would be reasonably likely to cause the failure of any condition precedent to Closing; or
- (Q) enter into any commitment or agreement to take any of the actions with respect to any of the foregoing.

7.2 Access for Due Diligence.

- (i) Subject to applicable Law, during the Interim Period, Seller shall (a) permit the Purchaser and its employees, agents, counsel, accountants or other representatives, to have reasonable access to (i) the premises of the Corporations, (ii) the Corporations' assets, including all Books and Records whether retained by Seller, the Corporations or otherwise, (iii) all Contracts, and (b) use commercially reasonable efforts to furnish to the Purchaser or its employees, agents, counsel, accountants or other representatives such material financial and operating data and other material information with respect to Corporations and its assets as the Purchaser from time to time reasonably requests.
- (ii) No investigations made by or on behalf of Purchaser, whether under this Section 7.2 or any other provision of this Agreement or any other Transaction Documents, will have the effect of waiving, diminishing the scope of, or otherwise affecting any representation or warranty made by Seller in this Agreement or any other Transaction Documents.

7.3 Sale of the PRV and PRV Reorganization

- (a) Seller shall work in good faith and use reasonable best efforts to sell the PRV as soon as commercially practicable, but in no event later than June 15, 2022 (the "PRV Sale Date"); provided that such period may be extended upon the mutual consent of the Parties or as may be necessary to comply with applicable law or orders of any Governmental Authority. Seller shall further keep Purchaser informed as to the status of sale activities and all offers for the PRV and provide copies thereof to the Purchaser, as well as copies of any contractual documentation relating to the sale.
- (b) It is understood among the Parties that, upon the closing of the sale of the PRV by PBT,
 - (i) Seller shall cause PBT, before the Second Closing, to pay to PBP an amount equal to thirty percent (30%) of the proceeds of the sale of the PRV, net of any reasonable and documented banker/broker commission and legal fees and the PRV Taxes in respect of such sales proceeds (the "Net Proceeds") plus \$5,000,000, pursuant to the Ryplazim MSA (the "Commitment Payments"); and

(ii)	pay the non-refundable balance of seventy percent (70%) (the "Seller PRV Percentage") of the
	Net Proceeds, less \$5,000,000, to Seller in repayment of the intercompany debt of PBT to Seller
	(the "Seller Debt Repayment"), as provided in the Pre-Closing Reorganization.

- (iii) Upon receipt of the payment described in 7.3(b)(ii), Seller shall contribute the remaining balance of the intercompany indebtedness owed to Seller by PBT to the capital of PBT and accordingly, the balance of the intercompany indebtedness would be cancelled, as provided in the Pre-Closing Reorganization (collectively the "PBT Reorganization").
- (c) In the event the Second Closing does not occur because the PRV has not been sold by the PRV Sale Date, notwithstanding anything in this Agreement to the contrary, to the extent the PRV is sold prior to the date which is three (3) months after the PRV Sale Date, the payment obligations of the Parties pursuant to Section 7.3(b) shall continue to apply, provided that (i) the "thirty percent (30%) of the proceeds of the sale of the PRV" under section 7.3.(b)(i) shall be replaced by "ten percent (10%) of the proceeds of the sale of the PRV" and (ii) the "non-refundable balance of seventy percent (70%)" under section 7.3.(b)(ii) shall be replaced by "non-refundable balance of ninety percent (90%)".
- (d) The Parties will reasonably cooperate to define the invoicing and payment conditions of the Commitment Payments and will agree to any reasonable amendments to the Ryplazim MSA that are required to permit such invoicing and payment to occur.
- (e) Seller shall determine the Net Proceeds and PRV Taxes before the Second Closing and provide Purchaser with a statement thereof. If the Parties are unable to agree on the Net Proceeds and the PRV Taxes, the matter will be submitted to an independent internationally recognized accounting firm for final determination, in accordance with the terms of Section 3.3, applicable *mutatis mutandis*. Such independent internationally recognized accounting firm shall, in the case of dispute, determine the PRV Taxes assuming that only those items of taxable income, deduction, loss, gain, credit or any other relevant item that are at least more likely than not required or permitted to be taken into account in accordance with applicable Tax Laws. The Parties agree that the PRV Taxes shall be determined on the basis that entire amount of the Commitment Payments is fully deductible for Tax purposes.

7.4 Pre-Closing Reorganization.

Notwithstanding anything to the contrary set forth herein, prior to the First Closing and Second Closing, as applicable, Seller shall cause the Corporations to effect and carry out the reorganization steps set forth in Schedule 7.4 (the "Pre-Closing Reorganization"); provided that Seller shall have the right to change the steps of the Pre-Closing Reorganization with the consent of the Purchaser (which shall not be unreasonably withheld). Such change in steps shall in no event involve or result in (i) PBT issuing shares or other equity or debt securities in exchange for the capitalization of its intercompany liabilities due to Seller as part of the Pre-Closing Reorganization or (ii) PBP adding to the stated capital of its shares an amount lower than the increase of the fair market value of the PBP shares resulting from the cancellation of the Intercompany liabilities due to Seller at the time of the First Closing; provided, further that the Purchaser shall have the right to be kept apprised of the implementation of such steps, shall have the right to receive any relevant documentation and provide any comments (which comments shall be taken into account by the Seller provided that they are reasonable and do not adversely impact the Seller).

7.5 Notice of Untrue Representation or Warranty.

Seller shall promptly notify Purchaser, and Purchaser shall promptly notify Seller, upon (i) any representation or warranty made by such Party contained in this Agreement becoming inaccurate during the Interim Period; or (ii) such Party's knowledge that any such representation or warranty was inaccurate when it was made on the date of this Agreement. For the purposes of this Section 7.5, each representation and warranty will be deemed to be given at and as of all times during the Interim Period. Any such notification must set out particulars of the untrue or incorrect representation or warranty and details of any actions being taken by the applicable Seller or Purchaser, as the case may be, to rectify that state of affairs.

7.6 Confidentiality.

Seller and the Purchaser acknowledges that a Confidential Disclosure Agreement dated September 30, 2020 was entered into between Kedrion and Seller (the "Confidentiality Agreement"). Each of the Seller, Kedrion and Purchaser agrees that the Confidentiality Agreement continues to apply and that each of the Seller, Kedrion and Purchaser is bound by its terms.

7.7 D&O Indemnification and Insurance.

- (i) Purchaser agrees that all rights to indemnification, advancement of expenses and exculpation from liabilities for acts or omissions occurring prior to the Closing Date existing in favour of the current or former directors or officers (each, a "D&O Indemnified Party") of each Corporation as provided in each of its respective Constating Documents or in any Contracts for indemnification in favour of such directors or officers (in each case in effect on the date hereof and made available prior to the date hereof to Purchaser) will remain obligations of each Corporation.
- (ii) After the Closing, Purchaser shall cause the Corporations to obtain and fully pay for customary, directors' and officers' and corporate liability insurance, fiduciary liability insurance and employment practices liability insurance covering the D&O Indemnified Parties providing benefits, terms, conditions, retentions in accordance with Kedrion Global Policy and levels of coverage that are at least as favourable to any beneficiary as such existing policies held by each member of the Corporations prior to the Closing.
- (iii) The provisions of this Section 7.7 are intended to be for the benefit of, and will be enforceable by, each D&O Indemnified Party and his or her heirs and representatives. From and after the Closing, Purchaser shall cause each Corporation to pay or cause to be paid (as incurred) all reasonable expenses, including reasonable fees and expenses of counsel, that a D&O Indemnified Party may incur in enforcing the indemnity and other obligations provided for in this Section 7.7 (subject to reimbursement (and a customary undertaking with respect thereto) if such D&O Indemnified Party is subsequently determined not to be entitled to indemnification under this Section 7.7).
- (iv) Purchaser shall cause the Corporations to obtain and fully pay for D&O insurance policy(ies) from a reputable insurer to come into effect at Closing.

7.8 Notice of Events; Updating Schedules.

During the Interim Period, Seller shall promptly notify the Purchaser in writing if the Seller becomes aware of (a) the occurrence or non-occurrence of any event or the existence of any fact or condition that would cause or constitute a breach of any of Seller's representations or warranties contained herein had such representation or warranty been made as of the time of such Seller's discovery of such event, fact or condition and (b) any failure on its part to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder.

7.9 Intellectual Property License.

Upon the First Closing until the Second Closing, Seller will cause PBT to provide [***] license to PBP to use all of the Corporations' Intellectual Property needed for the manufacturing of Ryplazim. The Parties agree that such license shall, in the event that the Second Closing does not occur and this Agreement is terminated in accordance with its terms, be terminated. The Parties agree and intend that a [***] license to use the Corporations' Intellectual Property needed for the manufacturing (including pursuant to a contract manufacturing or similar agreement) of Ryplazim be, and is, provided under the Ryplazim MSA and shall take any such steps required to amend the Ryplazim MSA in order to give effect to this Section 7.9. In the event the Second Closing has not occurred by [***], Seller shall, subject to payment of all regulatory and patent prosecution and annuities fees relating to Ryplazim, provide the PBP a [***] license for the manufacturing (including pursuant to a contract manufacturing or similar agreement), commercialization,

sale and distribution of Ryplazim to the extent necessary to permit PBP to conduct the business of the Ryplazim Targets and comply with its obligations under Section 8.6. Such license shall, in the event the Second Closing has not occurred by [***] at the latest, terminate on said date.

In the event the Seller has not sold the PRV and the Second Closing has not occurred by the PRV Sale Date, Purchaser shall purchase from Seller and Seller shall transfer, pursuant to the terms of an asset and technology transfer agreement to be entered into between the Parties on customary terms for [***], to Purchaser, or as designated by Purchaser, and the BLA and all Intellectual Property and Intellectual Property Rights relating to the Product (including any relevant dossiers, clinical studies, databases, records and the like), with such transfer to be effective no later than June 15, 2023.

7.10 Intercompany Indebtedness.

As soon as practicable following the date of this Agreement and in any event prior to the applicable Closing, Seller shall forgive and discharge any Indebtedness of the Ryplazim Targets held by Seller or its Affiliates (the "Intercompany Indebtedness") at no cost to the Ryplazim Targets.

7.11 Exclusivity.

During the period from the date of this Agreement until the earlier of the Second Closing or the termination of this Agreement in accordance with Article 11, Seller shall not, and shall cause the Ryplazim Targets not to, (a) solicit, encourage, seek, initiate, facilitate or engage in any discussions or negotiations with or (b) enter into any agreement with, any person or party (other than Purchaser) concerning (i) the purchase of any portion of the equity of, (ii) any merger or business combination with, or (iii) any sale of a majority of the consolidated assets (measured by fair market value) of, the Ryplazim Targets, and shall immediately cease any such discussions or negotiations that are ongoing and promptly request the return or destruction of any documents and information provided to any person or party in connection with such discussions.

7.12 Activities and Costs.

Effective as of the date hereof, Kedrion will be solely responsible for the costs and expenses until [***] for all reasonable fees and expenses, including reasonable allocations of resources available to PBP by Kedrion, as computed by Seller on the basis of supporting documentation to be provided to Kedrion, if required, to support continued operation of the plasma-derived business operations, to be paid in advance by Kedrion to Seller and not to exceed [***] per month. Such payments made by Kedrion to Seller shall be contributed by Seller to [***]. All payments made under this Section 7.12 shall be treated as an adjustment to the Purchase Price of the PBP Shares, as applicable, for all Tax purposes.

ARTICLE 8 POST-CLOSING COVENANTS AND AGREEMENTS

8.1 Books and Records.

The Seller shall deliver to the Purchaser on the Closing Date all books, papers, and records normally utilized in the operation of the Corporations. The Purchaser covenants and agrees that it will make available to the Seller at all times after the Closing Date upon reasonable notice any such books and records reasonably necessary to permit the Seller to prepare its tax returns, in connection with any claim under any insurance policies, in connection with any third-party claim, whether or not in litigation, in responding to any request by the FDA or other state, federal or international agency, or for any other reasonable purpose. The Purchaser shall retain such books and records consistent with the record retention policies of the Purchaser and applicable Laws.

113422456

8.2 Further Assurances.

- (a)General. The Parties agree that for the period after the Closing Date stipulated in the Transition Services Agreement, they will cooperate and use reasonable commercial efforts in order to perform such actions as may reasonably be required by the other Party for the purpose of assigning, transferring, granting, conveying, and confirming unto the Purchaser the ownership, possession or use of the Purchased Shares or otherwise confirming for the benefit of the Purchaser or the Seller any or all of the transactions contemplated by this Agreement including the ability to operate the Business in the ordinary course as set forth in the Transition Services Agreement. At any time or from time to time after the Closing, Seller shall, at the request of Purchaser and at no additional cost or expense, execute and deliver any further instruments or documents and take all such further action as Purchaser may reasonably request in order to evidence or effect the consummation of the transactions contemplated hereby. At any time or from time to time after the Closing, Purchaser shall, at the request of Seller and at no additional cost or expense, execute and deliver any further instruments or documents and take all such further action as Sellers may reasonably request in order to evidence or effect the consummation of the transactions contemplated hereby.
- (b)Wrong Pockets. If, at any time following the first Closing or the Second Closing, as applicable, any Party becomes aware that any asset or right (including with respect to the Non-Ryplazim Assets) that (a) should properly have been transferred to Purchaser such as to make true and correct the representation set forth in Section 5.5 and is held by Seller or an affiliate of Seller, or (b) is reasonably beneficial for the operation of the business of the Ryplazim Targets as conducted as of the date hereof or contemplated to be conducted after such Closing, then (i) Seller shall promptly transfer or cause its Affiliates to transfer such asset or right, as applicable, to Purchaser or as directed by Purchaser and (ii) Purchaser shall promptly assume or cause its Affiliates to assume such asset or right, in each case for no consideration and at the transferring Party's expense. Further, if at any time following the first Closing or the Second Closing, as applicable, Seller receives any notices or monies attributable to Ryplazim Targets or the business thereof, then Seller shall promptly deliver such notices or monies to or as directed by Purchaser. The Parties agree to treat the ultimate recipient of any such funds as having received the payment *ab initio* for all applicable Tax purposes.

8.3 Regulatory Notices

Each Party shall notify the appropriate Governmental Authorities and trade associations of the sale of the Purchased Shares including, but not limited to the FDA, any state, provincial or local registrations or licenses, if required by applicable laws or regulations.

8.4 Transfer Taxes.

All transfer, documentary, sales, use, stamp, registration and other such similar Taxes (other than income, franchise and similar Taxes) and fees (including any penalties and interest) incurred in connection with the transactions contemplated by this Agreement will be paid when due by the Party customarily responsible for same under Law, with the cost being borne by the Purchaser, and each Party, at its own expense, will file all necessary Tax Returns, reports and other documentation with respect to all such transfer, documentary, sales, use, stamp, registration and other taxes and fees, if required by Law with respect to the transactions contemplated herein. Each Party shall indemnify the other Party for all Damages incurred by such Party arising from a failure of the other Party to comply with the provisions of this Section, which obligation shall survive the Closing.

8.5 Consents, Notifications Waiting Periods.

In the event that any of transactions contemplated herein are subject to governmental and third party consents or notifications, or waiting periods that have not been procured or notice periods expired as of the Closing, the Parties shall use good faith efforts and cooperate with each other to promptly obtain the necessary authorizations, approvals, consents or waivers. Seller shall not be required to pay any consideration or compromise any rights not otherwise required by this Agreement to be compromised for

any such authorization, approval, consent or waiver, other than filing, recordation or similar fees, which shall be reimbursed by Purchaser.

8.6 Commercialization of Ryplazim and Compliance with PRV Legal Requirements

Purchaser shall cause PBT or an Affiliate thereof to commercialize Ryplazim within the time frame prescribed by the FDA in connection with the grant of the PRV and Purchaser shall not, or shall not cause the Corporations, to do or omit to do any act which act or omission would reasonably be expected to result in the termination, cancellation or revocation of the PRV. Kedrion shall cause the Corporations to comply with any responsibilities, obligations, requirements, parameters and conditions relating to the PRV set forth in the approval letter from the Department of Health and Human Services to Seller, Reference number 125659 regarding approval of the PRV or in any other correspondence received by the Corporations or its Affiliates from the FDA and shall not do or omit to do or cause any act which act or omission would reasonably be expected to result in the termination, cancellation, revocation or use of the PRV.

No later than [***], and subject to entering into license in Section 7.9 and Purchaser supplying the compassionate use patients, the Seller shall cause PBT to discontinue the expanded access for "compassionate use" of the Ryplazim product and shall cause any Ryplazim FFF product to be kept by PBT or, to the extent not held by PBT, transferred to PBT at no cost.

8.7 Non-Ryplazim Assets

Upon Purchaser's notice to Seller that Purchaser has determined not to acquire any Non-Ryplazim Assets, to be made no later than [***] prior to the Second Closing, Seller shall transfer or, with respect to any Non-Ryplazim Assets requiring the consent of non-affiliated third party to be so transferred, Seller shall use commercially reasonable efforts to transfer, all Non-Ryplazim Assets out of the Corporations prior to the Second Closing at no cost for the Corporations nor the Seller. The Non-Ryplazim Assets will be transferred to Seller in partial repayment of the debt due by the Corporations to the Seller, at fair market value thereof.

ARTICLE 9 CONDITIONS OF CLOSING

9.1 Conditions for the Benefit of Purchaser

The purchase and sale of the Purchased Shares is subject to the following conditions being satisfied on or prior to each Closing Date, which conditions are for the exclusive benefit of Purchaser and may be waived, in whole or in part, by Purchaser in writing in its sole discretion:

(a) Truth of Representations and Warranties and Performance of Covenants.

- (i) Each of the representations and warranties of Seller, except the Seller Core Representations which are the subject of Section 9.1(a)(ii) below, contained in this Agreement have been true and correct as of the date of this Agreement and will be true and correct in all material respects as of each Closing Date with the same force and effect as if such representations and warranties had been made on and as of each Closing Date;
- (ii) Each of the Seller Core Representations contained in this Agreement have been true and correct as of the date of this Agreement and will be true and correct in all respects as of each Closing Date with the same force and effect as if such representations and warranties had been made on and as each Closing Date;

- (iii) Seller shall have duly performed and complied with in all respects all of its material agreements and material covenants pursuant to this Agreement to the extent they are to be performed on or prior to each Closing Date;
- (iv) There shall be no statute, regulation, injunction, restraining order or decree or other governmental order of any nature of any court of competent jurisdiction or other Governmental Authority in effect that prohibits, enjoins or materially restrains the consummation of the transactions contemplated hereby; and
- (v) Since the date of this Agreement, there shall not have occurred a Material Adverse Effect.

and Seller shall have executed and delivered a certificate of a senior officer to that effect; and Purchaser will have received certificates confirming the above Sections 9.1(a)(i) and 9.1(a)(ii), signed for and on behalf of Seller by a senior officer or director of Seller (a "Seller Closing Certificate"). Upon the delivery of Seller Closing Certificate, the representations and warranties of the Seller in Article 5 and the Seller Core Representations will be deemed to have been made on and as of each Closing Date with the same force and effect as if made on and as of each Closing Date and as though the terms "Closing Date" and "the date of this Agreement" was substituted for the term, "the date hereof" or similar terms throughout such representations and warranties, except as would not have a Material Adverse Effect and except as may otherwise be qualified in Seller Closing Certificate.

- (b) **Deliveries.** The Seller shall have delivered or caused to be delivered to Purchaser the following for each Closing:
 - (i) share certificates representing the Purchased Shares duly endorsed in blank for transfer, or accompanied by irrevocable stock transfer powers duly executed in blank, in either case, by the holders of record, conveying to Purchaser all right, title and interest in the Purchased Shares, free and clear of all Encumbrances, together with evidence that Purchaser or its nominee(s) have been entered upon the books of the Corporations as the holder of the Purchased Shares;
 - (ii) certified copies of (i) the Constating Documents of the Corporations and (ii) all resolutions of the directors, and if required the shareholders, of Seller approving the entering into and completion of the transactions contemplated by this Agreement;
 - (iii) the Corporate Records;
 - (iv) a recent certificate of status, compliance, good standing or similar certificate with respect to each of the Corporations and Seller issued by the appropriate government officials of its jurisdiction of incorporation, and in the case of the Corporations, of each jurisdiction in which it carries on business:
 - (v) Seller Closing Certificate;
 - (vi) duly executed resignations and releases effective as at Closing of each director and officer of the Corporations specified by Purchaser in writing;
 - (vii) consents or releases in favour of the Corporations from the secured creditors identified in Schedule 9.1(b)(vii) with regard to assets of the Corporations which are subject to Encumbrances in their favour or an undertaking on their part that they will release any such Encumbrances promptly after Closing;

- (viii) a counterpart of the Transition Services Agreement duly executed by Seller (and/or applicable Corporation);
- (ix) a counterpart of the Statement of Work and related amendment to Ryplazim MSA duly executed by PBT;
- (x) a FIRPTA Certificate (in respect of the Second Closing); and
- (xi) a letter evidencing the consent or waiver of the change of control provision set forth in the Lease(s).

9.2 Conditions for the Benefit of Seller.

The purchase and sale of the Purchased Shares are subject to the following conditions to be fulfilled or performed at or before each Closing Date, which conditions are for the exclusive benefit of Seller and may only be waived, in whole or in part, by Seller in writing in its sole discretion:

- (a) Truth of Representations and Warranties and Performance of Covenants.
 - (i) Each of the representations and warranties of Purchaser, except the Purchaser Core Representations contained in this Agreement have been true and correct as of the date of this Agreement and will be true and correct in all material respects as of each Closing Date with the same force and effect as if such representations and warranties had been made on and as of each Closing Date;
 - (ii) Each of the Purchaser Core Representations contained in this Agreement have been true and correct as of the date of this Agreement and will be true and correct in all respects as of each Closing Date with the same force and effect as if such representations and warranties had been made on and as of each Closing Date;
 - (iii) Purchaser shall have duly performed and complied with in all material respects all of its agreements and covenants pursuant to this Agreement to the extent to be performed on or prior to each Closing Date; and
 - (iv) There shall be no statute, regulation, injunction, restraining order or decree or other governmental order of any nature of any court of competent jurisdiction or other Governmental Authority in effect that prohibits, enjoins or materially restrains the consummation of the transactions contemplated hereby.

and Purchaser shall have executed and delivered a certificate of a senior officer to that effect; and the Seller, on behalf of the Purchaser, will have received certificates confirming the above Sections 9.2(a)(i) and 9.2(a)(ii), signed for and on behalf of Purchaser by a senior officer or director of Purchaser (the "Purchaser Closing Certificate"). Upon the delivery of the Purchaser Closing Certificate, the representations and warranties of the Purchaser in Article 6 will be deemed to have been made on and as of each Closing Date with the same force and effect as if made on and as of the Closing Date and as though the terms "Closing Date" and "the date of this Agreement" was substituted for the term, "the date hereof" or similar terms throughout such representations and warranties, except as may be qualified in the Purchaser Closing Certificate.

- (b) **Deliveries.** Purchaser shall have delivered or caused to be delivered to Seller the following for each Closing:
 - (i) certified copies of the organizational documents of Purchaser and all resolutions of Purchaser's management committee and the sole member of Purchaser

approving the entering into and completion of the transactions contemplated by this Agreement;

- (ii) a recent certificate of status, compliance, good standing or similar certificate with respect to Purchaser issued by the appropriate government officials of its jurisdiction of incorporation;
- (iii) the Purchaser Closing Certificate;
- (iv) a counterpart of the Transition Services Agreement duly executed by Purchaser (and/or applicable Corporation);
- (v) a counterpart of the Statement of Work and related amendment to Ryplazim MSA duly executed by PBP; and
- (vi) insurance policies in effect prior to or at Closing as set forth in Section 7.7(iv).

9.3 Mutual Conditions.

The Second Closing is, subject to the following conditions, which condition is for the benefit of both Parties and may only be waived, in whole or in part, by both Parties in writing in their respective sole discretion: The Seller shall have completed or caused to be completed the sale of the PRV by PBT and the proceeds of such sale shall be fully distributed in accordance with Section 7.3 and the Pre-Closing Reorganization shall be completed to the satisfaction of both Parties in accordance with Schedule 7.4.

ARTICLE 10 CLOSING

10.1 Closing.

The completion of the transaction of purchase and sale pursuant to each Closing contemplated by this Agreement will take place on the third (3rd) Business Day after the date on which the last of the conditions to each Closing set forth in Article 9 (excluding conditions, that, by their terms, cannot be satisfied until the Closing) have been satisfied (or waived by the Party entitled to waive such condition) (the "**First Closing Date**", the "**Second Closing Date**", and each date, a "**Closing Date**"), or on such other date as the Parties may agree in writing.

10.2 Closing Procedures.

Subject to satisfaction or waiver by the relevant Party of the conditions of closing, on the Closing Date, Closing will take place remotely via the electronic exchange of documents and signatures between the Parties. On the Closing Date Seller shall deliver the Purchased Shares to the Purchaser and the Purchaser shall pay or satisfy the Purchase Price in accordance with Section 2.4.

ARTICLE 11 TERMINATION

11.1 Termination Rights.

This Agreement may, by notice in writing given prior to or on the Closing Date, be terminated:

- (i) by mutual consent of Purchaser and Seller;
- (ii) by Purchaser, if:

- (A) there has been a material breach of this Agreement by Seller that would give rise to the failure of any of the conditions specified in Section 9.1 and such breach has not been waived by Purchaser in writing or cured within 7 business days by Seller following written notice of such breach by Purchaser; or
- (B) if any of the conditions set forth in Section 9.1 have not been satisfied or waived on or prior to July 31, 2021 for the First Closing and the earlier of (1) June 15, 2022 or (2) thirty (30) days after the sale of the PRV, for the Second Closing; the Second Closing (each, an "Outside Date") and Purchaser has not waived such conditions in writing on or prior to the applicable Outside Date.
- (iii) by Seller, if:
 - (A) there has been a material breach of this Agreement by Purchaser that would give rise to the failure of any of the conditions specified in Section 9.2 and such breach has not been waived by Seller, in writing or cured within thirty (30) days by Purchaser following written notice of such breach by Seller.
 - (B) if any of the conditions set forth in Section 9.2 have not been satisfied or waived on or prior to the applicable Outside Date and Seller has not waived such conditions in writing on or prior to the applicable Outside Date.

11.2 Effect of Termination.

Upon termination of this Agreement pursuant to Section 11.1, this Agreement will forthwith become null and void, and have no further effect, the Parties will be released from all of their obligations under this Agreement and there will be no liability on the part of any Party, except for their respective obligations and liabilities under Section 7.6 (Confidentiality), this Section 11.2 (Effect of Termination), Section 15.3 (Public Announcements) and Section 15.1 (Fees and Expenses) which will survive any termination of this Agreement. In the event that either Party voluntarily breaches any material covenants of such Party under this Agreement, and, for reasons other than defaults caused by the other Party, third parties, facts and circumstances, or events out of the reasonable control of such Party, and such default cannot be cured and is not so cured, and as direct consequence thereof, the other Party terminates this Agreement and the Closing does not occur, such matter will survive the Termination and the breaching Party shall be liable to the other Party for damages arising out of such default.

ARTICLE 12 INDEMNIFICATION

12.1 Purchaser Indemnification.

From and after each Closing (but subject to the limitations set forth in Sections 12.3 and 12.4), Seller will indemnify and save harmless Purchaser and Kedrion, and each of their respective officers, directors, employees and agents (collectively, the "Purchaser Indemnified Parties" and each of them, a "Purchaser Indemnified Party"), from and against any and all Damages in respect of the applicable Corporation which any of Purchaser Indemnified Parties may suffer or incur as a result of, in respect of or arising out of:

(a) any failure of Seller to perform or fulfill any of its covenants under this Agreement for which a notice of claim under Section 12.5 has been provided to Seller within the applicable period specified in Section 12.3;

(b)	any breach, default or violation of any representation or warranty given by Seller set forth in Article 4 and
	Article 5 for which a notice of claim under Section 12.5 has been provided to Seller within the applicable
	period specified in Section 12.3;

- Damages or liabilities arising out of or resulting from (i) all unpaid Taxes, other than the PRV Taxes, with respect to all taxable years or periods of the applicable Corporation ending on or prior to the Closing which are not taken into account in the determination of Working Capital set forth in the applicable Closing Statement and any related interest, penalties or additions imposed thereon, (ii) all Taxes with respect to any breach of or inaccuracy in any representation or warranty of Seller contained in Section 5.1(r); (iii) any Taxes arising out of or attributable to an adjustment under Section 481(a) of the Code, and (iv) any Taxes attributable to receipt of indemnification pursuant to this Agreement, net of any Tax benefit derived in respect of the events or circumstances giving rise to such indemnification, in each case by the Purchaser Indemnified Parties;
- (d) any debts or liabilities whatsoever for Taxes, other than the PRV Taxes, of the applicable Corporation with respect to any and all Pre-Closing Tax Periods or the portion (the "Pre-Closing Portion") of any Straddle Period ending on the day before the applicable Closing Date, to the extent not taken into account in the applicable Closing Statement;
- (e) any debts or liabilities, other than the PRV Taxes, whatsoever relating to Pre-Closing Reorganization, to the extent not taken into account in the applicable Closing Statement;
- (f) an amount equal to the Seller PRV Percentage multiplied by the excess, if any, of (i) any income Taxes actually payable by PBT as finally determined by any applicable Taxation Authority as a result of the sale of the PRV over (ii) the PRV Taxes as determined before the Second Closing in accordance with Section 7.3, which such resulting amount may not, for the avoidance of doubt, be a negative number;
- (g) Any U.S. withholding taxes or other taxes arising from payment by PBT of its intercompany Liabilities;
- (h) the Seller Percentage of the PBP PRV Taxes, to the extent arising from a breach of, or material inaccuracy within, warranties under Section 5.1(r)(xix);
- (i) any Indebtedness to the extent not taken into account in the applicable Closing Statement; and
- (j) any Liabilities relating to or arising from the Pre-Closing Reorganization.

12.2 Seller Indemnification.

From and after each Closing (but subject to the limitations set forth in Sections 12.3 and 12.4), Purchaser will indemnify and hold harmless Seller and each of its respective officers, directors, employees, agents (collectively, "Seller Indemnified Parties" and each of them, a "Seller Indemnified Party") from and against all Damages in respect of the applicable Corporation which any of Seller Indemnified Parties may suffer or incur as a result of, in respect of or arising out of:

- (a) any failure of Purchaser to perform or fulfill any of its covenants under this Agreement for which a notice of claim under Section 12.5 has been provided to Purchaser within the applicable period specified in Section 12.3;
- (b) any breach, default or violation of any representation or warranty given by Purchaser set forth in Article 6 for which a notice of claim under Section 12.5 has been provided to Purchaser within the applicable period specified in Section 12.3;
- (c) an amount equal to the Seller PRV Percentage multiplied by the excess, if any, of (i) the PRV Taxes as determined before the Second Closing in accordance with Section 7.3 over

- (ii) any income Taxes actually payable by PBT as a result of the sale of the PRV, which such resulting amount may not, for the avoidance of doubt, be a negative number; and
- (d) any breach, default or violation of in relation to the PRV, including the obligation to perform its obligations under Section 8.6. For the avoidance of any doubt, the Parties agree that following the Second Closing, any obligations incumbent to PBT in relation to the PRV, including its commercialization within a specified timeframe, requested by the FDA, shall be fulfilled by PBT and Purchaser on a several and joint basis and Seller is entitled to be indemnified for actual damages incurred by Seller as a result of the Purchaser's non-fulfillment of any such obligation, including a failure to commercialize the PRV within the timeframe specified by the FDA.

12.3 Time Limitations.

- (a) The representations and warranties of Seller contained in this Agreement and the indemnification obligations of the Seller pursuant to Section 12 will survive each Closing and, notwithstanding the Closing, continue in full force and effect for a period of [***], except that:
 - (i) the Seller Core Representations will survive Closing and continue in full force and effect [***];
 - (ii) the representations and warranties set out in Section 5.1(r) (Tax Matters) or indemnification in Sections 12.1(f), 12.1(g), 12.1(h) and 12.1(j) will survive until the date that is [***] following the expiration of the period ("Tax Assessment Period") during which any Tax assessment may be issued by a Taxation Authority in respect of any matters to which such representations and warranties extend. The Tax Assessment Period will be determined having regard to any consent, waiver, agreement or other document that extends the period during which a Taxation Authority may issue a Tax assessment granted with the prior consent of Seller (not to be unreasonably withheld). A Tax assessment includes any assessment, reassessment or other form of recognized document assessing liability for Taxes under applicable Law;
 - (iii) any Claim involving fraud or fraudulent misrepresentation, intentional fault of Seller will survive Closing and continue in full force and effect [***]; and
 - (iv) any claim brought to Seller prior to the date which is [***] after the Closing shall survive until final resolution thereof.
- (b) The representations and warranties of Purchaser will survive Closing and, notwithstanding the Closing, continue in full force and effect for a period of [***] after the Closing except that:
 - (i) the Purchaser Core Representations will survive Closing and continue in full force and effect [***];
 - (ii) any Claim involving fraud, fraudulent misrepresentation, intentional fault of Purchaser will survive Closing and continue in full force and effect [***].

12.4 Limitations on Indemnification.

(a) Neither Party has any liability or obligation with respect to any incidental, consequential, special and punitive damages or lost profits arising out of, resulting from or relating to a breach of this Agreement.

113422456

- (b) Seller have no liability for, or obligation with respect to, any claims for indemnification for environmental matters under Section 12.1 (the "Environmental Claims"), other than the representations and warranties in Section 5.1(m), arising from any change in the use of the Leases or cessation of activities threat. Seller has no liability with respect to any Environmental Claims: (i) in response to a claim asserted by a Governmental Authority or other third party which relates to matters, facts and circumstances occurring or existing prior to Closing, for any Damages for actions that are not required by the Governmental Authority or third party; (ii) for any Damages which result from Purchaser conducting invasive investigations, sampling or monitoring of the premises subject to the Leases or disclosing information to Governmental Authority or third parties from and after Closing except, and without limiting (i) above, (A) as required by Law, (B) as required by an order of a court of competent jurisdiction or (C) as a result of a Person, who is dealing at arms' length with Purchaser and who has signed a letter of intent to purchase at least a controlling interest in the business of the Corporations (either through assets or shares), conducting testing; or (iii) for any Damages, to the extent they were exacerbated or allowed to be exacerbated by Purchaser.
- (c) The indemnification obligations of Seller pursuant to Section 12.1 will:
 - (i) only apply to aggregate Damages with respect to all indemnification claims pursuant to Section 12.1 which exceed [***], upon which the indemnification obligations of Seller only applies to the amount by which such aggregate Damages exceeds [***]; and
 - (ii) only apply to any individual claim for Damages with respect to indemnification pursuant to Section 12.1 if the aggregate amount of the Damages with respect to such claim, or a series of claims arising out of a related set of facts, circumstances or occurrences, is greater than [***]. The amount of Damages with respect to any individual claim that is less than or equal to [***] is not taken into account in determining whether or not the total of all Damages meets the threshold amount under Section 12.4(c)(i); and
 - (iii) not exceed [***] in the aggregate.
- (d) Section 12.4(c) does not apply to indemnification claims with respect to:
 - (i) the Seller Core Representations, the representations and warranties set out in Section 5.1(r) (Tax Matters), or claims based on fraud, fraudulent or willful misconduct, or intentional or gross fault of Seller. The indemnification obligations of Seller for claims with respect to Seller Core Representations and with respect to Section 5.1(r) (Tax Matters) will not exceed [***] in the aggregate with Section 12.1(c); and
 - the indemnification obligations of Seller for claims relating to Tax matters in Sections 12.1(f), 12.1(g) and 12.1(h)) ("Other Tax Matters") will not exceed [***] in the aggregate with Section 12.1(c) and 12.1(d)(i); provided that the increased Indemnification obligations of Seller relating to Other Tax Matters shall only apply in the event that Purchaser's or the Corporations' Damages in respect thereof exceed [***], and in such event, only apply to the amount by which such aggregate Damages exceed [***].
- (e) Payments by any Indemnifying Party pursuant to Section 12.1 or Section 12.2 in respect of any Damages will be reduced dollar-for-dollar by:

- (i) the amount by which the Tax liability of the Indemnified Party is reduced as a result of such Damages including without limitation, by deduction, reduction of income by virtue of tax basis or otherwise to the extent that such reduction is or will be realized (including through a carryback to any of the three previous taxable periods) in the same taxable period as the date of the indemnity payment or any of the five subsequent calendar years;
- (ii) the amount of any cash-funded provision or reserve for the matters giving rise to such Damages that is identified for such matters in the Financial Statements or Interim Financial Statements or in the Closing Statement; and
- (iii) the net amounts (less any deductible, premium increase, defense costs or other costs or outof-pocket expenses) actually recovered by the Indemnified Party under any insurance policies, indemnity or reimbursement Contract or any claim or settlement with any other Person with respect to such Damages;
- (f) Notwithstanding anything to the contrary in this Agreement, the Purchaser Indemnified Parties shall not have any right to indemnification under this Agreement with respect to Taxes to the extent such Taxes: (a) result from any transactions undertaken or actions taken by the Purchaser or any of its Affiliates (including any of the Corporations) after the applicable Closing, without the consent of the Seller, including, for greater certainty, any amendment, refiling or other modification of any Tax Return of the applicable Corporation with respect to any period ending on or prior to the Closing Date or the initiation of any voluntary disclosure process or other similar discussions with applicable Taxation Authorities in respect of any such period; (b) result from or arise in connection with a breach or non-fulfillment of any representation, warranty or covenant contained in this Agreement by the Purchaser or any of its Affiliates; or (c) relate to any Tax period (or portion thereof) ending after the applicable Closing, including, for greater certainty, in respect of the unavailability in any such period of any Tax losses, credits, expenses, deductions, cost basis or any other Tax attribute.
- (g) For the purposes of Section 12.1(c), the portion of Taxes allocable to the Pre-Closing Portion of the Straddle Period of PBT shall be (i) in the case of Taxes, other than Taxes based upon or related to income or receipts, or franchise Taxes, or Taxes based on capitalization, debt or shares of stock authorized, issued or outstanding, or ad valorem Taxes, be deemed to be the amount of such Tax for the entire taxable period multiplied by a fraction the numerator of which is the number of days in the Straddle Period ending on the day before the Closing Date and the denominator of which is the number of days in the entire Straddle Period, and (ii) in the case of any Tax based upon or related to income or receipts, or franchise Taxes, or Taxes based on capitalization, debt or shares of stock authorized, issued or outstanding, or ad valorem Taxes, be deemed equal to the amount which would be payable if the relevant Straddle Period ended as of the day immediately prior to the Closing Date.

12.5 Notification.

Promptly upon obtaining knowledge thereof, the party being indemnified under this Agreement (the "Indemnified Party") shall notify the party providing for indemnification under this Agreement (the "Indemnifying Party"), as the case may be, in writing of any cause which entitles the Indemnified Party to make a claim for indemnification under this Article 12 (an "Indemnification Notice").

12.6 Direct Claim

(a) Following receipt of Indemnification Notice pursuant to Section 12.5 relating to a claim that is not a Third Party Claim (a "**Direct Claim**"), the Indemnifying Party, after giving notice to the Indemnified Party within twenty (20) days of such receipt, has thirty (30) days to

investigate the Direct Claim and respond in writing. For purposes of the investigation, the Indemnified Party shall make available to the Indemnifying Party the information relied upon by the Indemnified Party to substantiate the Direct Claim, together with such other information as the Indemnifying Party may reasonably request.

(b) If, after the investigation, the Indemnifying Party disputes the validity or amount of the Direct Claim, the Indemnifying Party shall provide written notice of the dispute to the Indemnified Party within the thirty (30) day period specified in Section 12.6(a). The dispute notice will describe in reasonable detail the nature of the Indemnifying Party's dispute. During the thirty (30) day period immediately following receipt of a dispute notice by the Indemnified Party, the Indemnifying Party and the Indemnified Party shall attempt in good faith to resolve the dispute. If the Indemnifying Party and the Indemnified Party fail to resolve the dispute within that thirty (30) day time period, the Indemnified Party is free to pursue all rights and remedies available to it, subject only to this Agreement, including its right to submit such dispute to a court of competent jurisdiction in accordance with Section 15.11. If the Indemnifying Party fails to respond in writing to the Direct Claim within the thirty (30) day period specified in Section 12.6(a), the Indemnifying Party is deemed to have rejected the Direct Claim in which event the Indemnified Party is free to pursue all rights remedies available to it, subject to this Agreement.

12.7 Defense of Third Party Claim.

- (a) If any legal proceeding shall be instituted or any Claim or demand shall be asserted by a third party against the Indemnified Party (each a "Third Party Claim"), then the Indemnifying Party shall have the right, after receipt of the Indemnification Notice under Section 12.5 and upon giving notice to the Indemnified Party within fifteen (15) days of such receipt, to participate in the investigation and defense of the Third Party Claim and may also elect to assume the investigation and defense of the Third Party Claim.
- (b) If the Indemnifying Party assumes the investigation and defense of a Third Party Claim:
 - (i) the Indemnifying Party shall pay for all costs and expenses of the investigation and defense of the Third Party Claim except that the Indemnifying Party will not, so long as it diligently conducts such defense, be liable to the Indemnified Party for any fees of other counsel or any other expenses with respect to the defense of the Third Party Claim, incurred by the Indemnified Party after the date the Indemnifying Party validly exercised its right to assume the investigation and defense of the Third Party Claim; and
 - (ii) the Indemnifying Party shall reimburse the Indemnified Party for all reasonable costs and expenses incurred by the Indemnified Party in connection with the investigation and defense of the Third Party Claim prior to the date the Indemnifying Party validly exercised its right to assume the investigation and defense of the Third Party Claim.
- (c) If the Indemnified Party undertakes the defense of the Third Party Claim, the Indemnifying Party will not be bound by any determination of the Third Party Claim or any compromise or settlement of the Third Party Claim effected without the consent of the Indemnifying Party (which consent may not be unreasonably withheld or delayed).
- (d) The Indemnifying Party will not be permitted to compromise and settle or to cause a compromise and settlement of any Third Party Claim without the prior consent of the Indemnified Party (which consent may not be unreasonably withheld or delayed), unless:

- (i) the terms of the compromise and settlement require only the payment of money for which the Indemnified Party is entitled to indemnification under this Agreement;
- (ii) the terms of the compromise and settlement do not require the Indemnified Party to admit any wrongdoing, take or refrain from taking any action, acknowledge any rights of the Person making the Third Party Claim or waive any rights that the Indemnified Party may have against the Person making the Third Party Claim; and
- (iii) the Indemnified Party receives, as part of the compromise and settlement, a legally binding and enforceable unconditional release from any and all obligations or liabilities it may have with respect to the Third Party Claim.
- (e) If the Indemnifying Party assumes the investigation and defense of the Third Party Claim, the Indemnified Party shall cooperate on a timely basis with the Indemnifying Party in the defense of the Third Party Claim and shall use its commercially reasonable efforts to make available to the Indemnifying Party or its representatives on a timely basis, at the request and expense of the Indemnifying Party:
 - (i) those employees whose assistance, testimony or presence is necessary to assist the Indemnifying Party in investigating and defending the Third Party Claim; and
 - (ii) all documents, records and other materials in the possession, control or power of the Indemnified Party, reasonably required by the Indemnifying Party for its use solely in defending the Third Party Claim.
- (f) The Indemnified Party and the Indemnifying Party shall keep each other fully informed of the status of any Third Party Claim and any related proceedings.
- (g) In the case of any conflict between this Section 12.7 and the provisions of Section 13.3 with respect to a Tax Matter, Section 13.3 shall govern.

12.8 Exclusion of Other Remedies

Except as provided in this Section 12.8, following Closing, the indemnities provided in Section 12.1 and Section 12.2 constitute the only remedy of Seller or Purchaser, respectively, against a Party in the event of any breach of a representation, warranty, covenant or agreement of such Party contained in this Agreement. Prior to Closing, the Parties may exercise their rights of termination in Section 11.1, as applicable. Accordingly, a Party may seek to enforce the performance of this Agreement by injunction or specific performance upon application to a court of competent jurisdiction without proof of actual Damages (and without requirement of posting a bond or other security). The Parties expressly waive and renounce any other remedies whatsoever which it would otherwise be entitled to as against any other Party.

12.9 One Recovery

Any Indemnified Party is not entitled to double recovery for any claims even though they may have resulted from the breach of more than one of the representations, warranties, covenants and obligations of the Indemnifying Party in this Agreement. No Party has any liability or obligation with respect to any claim for indemnification to the extent that such matter was reflected as an adjustment to the Purchase Price in Article 3.

12.10 Duty to Mitigate

Nothing in this Agreement in any way restricts or limits the general obligation at Law of an Indemnified Party to mitigate any loss which it may suffer or incur by reason of the breach by an Indemnifying Party of any representation, warranty, covenant or obligation of the Indemnifying Party under

this Agreement. If any claim can be reduced by any recovery, settlement or otherwise under or pursuant to any insurance coverage, indemnity or reimbursement agreement or Contract, or pursuant to any claim, recovery, settlement against or with any other Person, the Indemnified Party shall take all appropriate steps to enforce such recovery, settlement or payment.

12.11 Adjustment to Purchase Price

Any payment made by Seller pursuant to this Article 12 will constitute a dollar for dollar decrease of the Purchase Price and any payment made by Purchaser pursuant to this Article 12 will constitute a dollar for dollar increase in the Purchase Price.

ARTICLE 13 TAX MATTERS

13.1 Tax Returns.

- (a) Seller shall prepare, or cause to be prepared, all income Tax Returns ("Pre-Closing Returns") each of the Corporations for Tax periods ending on or prior to the applicable Closing Date that are due after the applicable Closing Date. Each such Pre-Closing Return shall be prepared in accordance with existing procedures, practices and accounting methods of the applicable Corporation, unless such procedure, practice, accounting method or other contemplated treatment is not permitted under Law. Seller shall provide Purchaser with a draft of any such Pre-Closing Return no later than 30 days before the filing due date for such Pre-Closing Return, and Seller shall make all reasonable changes to such Pre-Closing Return requested by Purchaser within 15 days after receipt by Purchaser of such draft Pre-Closing Return. Notwithstanding the foregoing, the Parties agree that, in connection with such Pre-Closing Returns, Seller shall have sole discretion to cause PBP to make any (i) designation pursuant to subsections 80 (5) (10) of the Tax Act (ii) agreement pursuant to section 80.04 of the Tax Act, and (iii) election pursuant to subsection 256(9) of the Tax Act (each, a "Specified Tax Position"). Purchaser shall cause each of the Corporations to timely file each such Pre-Closing Return prepared in accordance with the foregoing.
- Seller may prepare, or cause to be prepared, amended income Tax Returns ("Amended Pre-Closing Returns") of each of the Corporations for Tax periods ending on or prior to the applicable Closing Date. Each such Amended Pre-Closing Return shall be prepared as determined by Seller and in accordance with existing procedures, practices and accounting methods of the applicable Corporation, unless such procedure, practice, accounting method or other contemplated treatment is not permitted under Law. Seller shall provide Purchaser with a draft of any such Amended Pre-Closing Return no later than 30 days before the filing due date for such Amended Pre-Closing Return, and Seller shall make all reasonable changes to such Amended Pre-Closing Return requested by Purchaser within 15 days after receipt by Purchaser of such draft Amended Pre-Closing Return. Seller shall have sole discretion to cause the Corporations to make, amend or modify any Specified Tax Position (including in connection with any Tax Matter). Purchaser shall cause the Corporations to timely file each such Amended Pre-Closing Return prepared in accordance with the foregoing.
- (c) Purchaser shall prepare, or cause to be prepared, all income Tax Returns ("Straddle Returns") of PBT for Tax periods beginning before and ending after the applicable Closing Date. Each such Straddle Return shall be prepared in accordance with existing procedures, practices and accounting methods of PBT, unless such procedure, practice, accounting method or other contemplated treatment is not permitted under Law. Purchaser shall provide Seller with a draft of any such Straddle Return no later than 30 days before the filing due date for such Straddle Return, and Purchaser shall make all reasonable changes to such Straddle Return requested by Seller within 15 days after receipt by Seller of such

draft Straddle Return to the extent they relate to a portion of any Straddle Period up to and including the Closing. Purchaser shall cause PBT to timely file each such Tax Return prepared in accordance with the foregoing.

- (d) Purchaser shall and shall cause the Corporations to fully cooperate with Seller (including allowing access by Seller and its representatives to the books and records of the Corporations) in connection with the preparation or review of any Tax Returns in accordance with Sections 13.1(a), 13.1(a) and 13.1(c).
- (e) Without the prior written consent of Seller, Purchaser shall not, and shall not cause or permit any Corporation, to amend, refile or otherwise modify any Tax Return of a Corporation with respect to any Tax period beginning prior to the applicable Closing, if such amendment, refiling or modification would result in an increase in the amount of the Taxes for which Seller may be liable under this Agreement.
- (f) The parties shall act in good faith to resolve any dispute prior to the date on which a disputed Straddle Return is required to be filed. If the parties cannot resolve any disputed item, the item in question shall be referred to the Independent Accountant for final resolution, in accordance with the standards and procedures set forth in Section 3.3(a), *mutatis-mutandis*. No later than five (5) days following the final resolution and filing of such Straddle Return, Seller shall pay Purchaser (in immediately available funds) the amount of Taxes in respect of which Seller would owe an indemnification obligation pursuant to Section 12.1(c). Subject to the preceding sentence, Purchaser shall pay or cause to be paid all Taxes due and payable in respect of all such Straddle Returns.

13.2 Tax Cooperation.

Seller and Purchaser agree to furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information (including access to books and records and individuals) and assistance relating to the Corporations as is reasonably requested for the filing of any Tax Returns, Tax certificates and any other filings in connection with the transactions contemplated by this Agreement or otherwise and for the preparation of any Tax Matter and for the prosecution or defense of any Tax claim. Seller and Purchaser agree that each shall preserve and keep all books and records with respect to Taxes and Tax Returns in such party's possession as of the applicable Closing Date, or as later come into such party's possession, until the expiration of the applicable statute of limitations. Any information obtained under this Section 13.3 shall be kept confidential except (i) as may be otherwise necessary in connection with the filing of Tax Returns or claims for refund or in conducting a Tax Matter or other proceeding or (ii) with the consent of Seller or Purchaser, as the case may be.

13.3 Conduct of Tax Matters.

- (a) If: (i) a Taxation Authority asserts any actions, inquiries, claims, assessments, audits or similar events with respect to Taxes against either of the Corporations, and (ii) the Seller could be responsible for any portion of the related Taxes (any such claim, a "Tax Matter"), then the party first receiving notice (whether directly, or indirectly through an Affiliate of such party) of such Tax Matter shall promptly provide to the Purchaser, the applicable Corporation and the Seller written notice specifying in reasonable detail the basis for such Tax Matter and shall include a copy of the relevant correspondence received from the Taxation Authority in respect of such Tax Matter; provided, however, that a failure by either party to give such notice shall not affect the other party's rights to indemnification under Article 12 unless and solely to the extent such other party is materially and adversely prejudiced as a consequence of such failure.
- (b) If, within thirty days after the Seller receives notice of a Tax Matter, the Seller provides to the Purchaser a written notice in which the Seller elects to contest, and to control the defense or prosecution of, such Tax Matter, then subject to the provisions of this Section

13.3(b), the Seller shall have the right to defend or prosecute and the right to control, at its sole cost and expense, such Tax Matter by all appropriate proceedings; provided that, notwithstanding the foregoing, the Seller and Purchaser shall jointly control any such Tax Matter that would reasonably be expected to have a material adverse impact on the Taxes of the Purchaser or the applicable Corporation with respect to a taxable period (or portion thereof) beginning on or after the Closing Date (a "Joint Controlled Proceeding"). For any such Tax Matter, the defense or prosecution of which the Seller controls (a "Seller Controlled Proceeding"): (i) the Seller shall defend or prosecute the Tax Matter diligently and in good faith; (ii) the Seller shall not, without the prior written consent of the Purchaser, which consent shall not be unreasonably withheld, conditioned or delayed, enter into any compromise or settlement of such Tax Matter; (iii) the Seller shall inform the Purchaser of all material developments and events relating to such Tax Matter (including providing to the Purchaser copies of relevant portions of all written materials relating to such Tax Matter); (iv) the parties shall cooperate with each other and each party's authorized representatives in good faith in order to contest effectively such Tax Matter; (v) the Purchaser or its authorized representative shall be entitled, at the expense of the Purchaser, to attend and participate in all conferences, meetings and proceedings relating to such Tax Matter; and (vi) if the Purchaser reasonably withholds consent to a compromise or settlement pursuant to subsection (ii), above, then the Purchaser shall be entitled to participate with the Seller in joint control of the defense or prosecution of such Tax Matter (which shall thereafter be a Joint Controlled Proceeding).

- (c) The Seller's right to control at any time any Tax Matter pursuant to Section 13.3(b) shall only apply after payment to the relevant Corporation of the amount of Tax subject at that time to collection measures by the applicable Governmental Authority (the "Prepaid Amount") within the time prescribed by Law. In the event such Tax Matter is subsequently settled in accordance with the terms of a settlement or final non-appealable judgment, the applicable Corporation shall promptly reimburse to the Seller the lesser of: (i) the Prepaid Amount and any interest received from the Governmental Authority thereon (net after-Tax); and (ii) the Prepaid Amount and any interest received from the Governmental Authority thereon (net after-Tax) less the final amount of Tax due pursuant to such Tax Matter in accordance with the terms of such settlement or final non appealable judgment.
- (d) If the Seller fails within thirty (30) days after receipt of notice to assume control of a Tax Matter (other than a Joint Controlled Proceeding) in accordance with Section 13.3(b), or fails to pay the Prepaid Amount to the relevant Corporation in accordance with Section 13.3(c), then Purchaser will provide to Seller a reasonable opportunity to comment on any representations or submissions proposed to be made to a Governmental Authority in respect of such Tax Matter and to attend any meeting with any such Governmental Authority with respect to such Tax Matter, and Purchaser shall not settle, resolve or abandon (and shall not allow the Corporations to settle, resolve or abandon) such Tax Matter without the prior written consent of Purchaser (which shall not be unreasonably withheld, conditioned or delayed).

13.4 Conflicts; Survival.

In the event of a conflict between this Article 13 and any other provision of this Agreement, this Article 13 shall govern and control. This Article 13 shall survive until the expiration of the applicable statute of limitations (taking into account all extensions thereof granted with the prior written consent of the Seller, not to be unreasonably withheld).

13.5 Excessive Dividend Elections.

(a) If a Taxation Authority determines that PBP has made an "excessive eligible dividend designation" (as defined in subsection 89(1) of the Tax Act) in respect of any dividend paid before Closing, the Seller hereby irrevocably concurs (or shall cause the recipient of the

relevant dividend or any successor thereof to concur) in the making of an election under subsection 185.1(2) of the Tax Act in respect of the full amount thereof, and such election shall be made by PBP in the manner and within the time prescribed by subsections 185.1(2) and 185.1(3) of the Tax Act. The Seller further irrevocably agrees to do all such things and take all such actions as the Purchaser may reasonably request to ensure that PBP does not incur any liability for making an "excessive eligible dividend designation".

13.6 §338(g)Election.

Purchaser shall pay any state or local U.S. Tax and indemnify Seller against all such Taxes and any other Damages attributable to the making by Purchaser of an election under Code §338(g) (and any corresponding elections under state or local U.S. law) with respect to the purchase and sale of the Purchased Shares of PBP.

ARTICLE 14 OFAC; ANTI-MONEY LAUNDERING; PATRIOT ACT.

14.1 OFAC Representation.

Seller and Purchaser hereby represent, warrant and certify to and for the benefit of the other that such Party is not now, and has never been, nor shall it be, at any time prior to the Closing, an individual, corporation, partnership, limited partnership, joint venture, association, joint stock company, trust, trustee, estate, limited liability company, unincorporated organization, real estate investment trust, government or any agency or political subdivision thereof, or any other form of entity with whom a United States citizen, entity organized under the laws of the United States or its territories or entity having its principal place of business within the United States or any of its territories, is prohibited from transacting business of the type contemplated by this Agreement, whether such prohibition arises under United States law, regulation, executive orders and lists published by the Office of Foreign Assets Control, Department of the Treasury ("OFAC") (including those executive orders and lists published by OFAC with respect to Persons that have been designated by executive order or by the sanction regulations of OFAC as Persons with whom U.S. Persons may not transact business or must limit their interactions to types approved by OFAC ("Specially Designated Nationals and Blocked Persons")) or otherwise. No party or any Person who owns an interest in such party is now or has ever been, nor shall be at any time prior to the Closing, a Person with whom a U.S. Person, including a "financial institution" as defined in 31 U.S.C. 5312 (a)(z), as periodically amended ("Financial Institution"), is prohibited from transacting business of the type contemplated by this Agreement, whether such prohibition arises under United States law, regulation, executive orders and lists published by the OFAC (including those executive orders and lists published by OFAC with respect to Specially Designated Nationals and Blocked Persons) or otherwise.

14.2 Purchaser's Assurances.

Purchaser represents, warrants and certifies to and for the benefit of Seller that it has taken, and shall continue to take until the Closing or any earlier termination of this Agreement, such measures as are required by applicable Law to assure that the funds used to pay all or any portion of the Purchase Price originate: (i) from transactions that do not violate United States law nor, to the extent such funds originate outside the United States, do not violate the laws of the jurisdiction in which they originated; and (ii) from permissible sources under United States law and, to the extent such funds originate outside the United States, under the laws of the jurisdiction in which they originated.

14.3 Anti-Money Laundering.

Seller represents, warrants and certifies to and for the benefit of Purchaser that neither Seller nor either of the Corporations: (i) is under investigation by any governmental authority for, or has been charged with, or convicted of, money laundering, drug trafficking, terrorist-related activities, any crimes which in the United States would be predicate crimes to money laundering, or any violation of any Anti-Money

113422456

Laundering Laws (as hereinafter defined); (ii) has been assessed civil or criminal penalties under any Anti-Money Laundering Laws; or (iii) has had any of its funds seized or forfeited in any action under any Anti Money Laundering Laws. For purposes of subclause (i) of the previous sentence, the term "Anti-Money Laundering Laws" shall mean all applicable laws, regulations and sanctions, state and federal, criminal and civil, that: (w) limit the use of and/or seek the forfeiture of proceeds from illegal transactions; (x) limit commercial transactions with designated countries or individuals believed to be terrorists, narcotics dealers or otherwise engaged in activities contrary to the interests of the United States; (y) require identification and documentation of the parties with whom a Financial Institution conducts business; or (z) are designed to disrupt the flow of funds to terrorist organizations. Such laws, regulations and sanctions shall be deemed to include, without limitation, the USA PATRIOT Act of 2001, Pub. L. No. 107-56 (the "Patriot Act"), the Bank Secrecy Act of 1970, as amended, 31 U.S.C. Section 5311 et seq., the Trading with the Enemy Act, 50 U.S.C. App. Section 1 et seq., the International Emergency Economic Powers Act, 50 U.S.C. Section 1701 et seq., and the sanction regulations promulgated pursuant thereto by the OFAC, as well as laws relating to prevention and detection of money laundering in 18 U.S.C. Sections 1956 and 1957. The parties represent, warrant and certify to and for the benefit of the other party that they are in compliance with any and all applicable provisions of the Patriot Act.

ARTICLE 15 GENERAL

15.1 Fees and Expenses.

Except for the fees and expenses related to the 2019 valuation of the PRV which shall borne equally by both Purchaser and Seller, Purchaser will pay for its own fees and expenses and Seller shall pay for its own fees and expenses which are, in each case, incurred in connection with the negotiation, preparation, execution and performance of this Agreement, the Transaction and the agreements contemplated by them, including the fees and expenses of legal counsel, investment advisers and accountants.

15.2 Further Assurances.

Each Party will, from time to time, execute and deliver all such further documents and instruments and do all acts and things as the other Party may reasonably require to effectively carry out or better evidence or perfect the full intent and meaning of this Agreement.

15.3 Public Announcements.

The Purchaser nor any of its directors, officers, employees and respective representatives will make any public disclosure concerning the matters set forth in this Agreement or the transactions contemplated by this Agreement without the prior written consent of the Seller, except if required by Law or rules of applicable stock exchanges in which case the Purchaser shall use commercially reasonable efforts to give the Seller an opportunity to review and comment on any such disclosure in advance of public release.

15.4 Benefit of the Agreement.

This Agreement will inure to the benefit of and be binding upon the respective heirs, executors, administrators, other legal representatives, successors and permitted assigns of the Parties.

15.5 Entire Agreement.

This Agreement and the other Transaction Documents constitute the entire agreement between the Parties with respect to the respective subject matters hereof and thereof, and cancel and supersede any prior understandings and agreements between the Parties with respect thereto. There are no representations, warranties, terms, conditions, undertakings or collateral agreements, express, implied or statutory, among the Parties other than as expressly set forth in this Agreement. In the event of any

113422456

inconsistency between the statements in the body of this Agreement and the Schedules hereto (other than an exception expressly set forth as such in a Schedule hereto), the statements in the body of this Agreement will control. The Option Agreement is hereby superseded and terminated, and shall be of no further force or effect.

15.6 Amendments and Waivers.

No amendment to this Agreement will be valid or binding unless set forth in writing and duly executed by Seller and Purchaser. Except as expressly provide in this Agreement, no waiver of any breach of any provision of this Agreement will be effective or binding unless made in writing and signed by the Party purporting to give the same and, unless otherwise provided, will be limited to the specific breach waived. A Party's failure or delay in exercising any right under this Agreement will not operate as a waiver of that right. A single or partial exercise of any right will not preclude a Party from any other or further exercise of that right or the exercise of any other right.

15.7 Assignment.

No Party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of the other party hereto, which consent will not be unreasonably withheld or delayed; *provided* that Purchaser shall be permitted to assign, delegate or otherwise transfer its rights, interest and obligations under this Agreement (including prior to the consummation thereof) to (a) any direct or indirect wholly-owned subsidiary of Purchaser or to any Affiliate of which Purchaser is a direct or indirect wholly-owned subsidiary (provided that Purchaser remains solidarily bound by the terms and conditions of this Agreement so assigned, including for all its obligations hereunder) or (b) in connection with the transfer by Purchaser of all or substantially all of the capital stock and/or assets of the Company and or its Subsidiaries; *provided* that until the date which is twelve (12) months after the date of this Agreement, any such transfer as provided in (b) shall require the consent of Seller, which shall not be unreasonably conditioned, withheld or delayed.

15.8 Time of Essence.

To Purchaser:

Time is of the essence of this Agreement. The mere lapse of time in the performance of the terms of this Agreement by any Party will have the effect of putting such Party in default.

15.9 Notices.

Any demand, notice or other communication to be given in connection with this Agreement must be given in writing and will be given by personal delivery or by facsimile or email (where applicable) addressed to the recipient as follows:

```
Kedrion SPA
Sede Legale – Loc. Ai Conti,
55051 Castelvecchio Pascoli – Barga (LU)
with a copy to (which will not constitute notice):
Kedrion Biopharma Inc. Parker Plaza – 400 Kelby Street
Fort Lee, New Jersey 07027
Attention: [***]
Email: [***]
```

White & Case LLP Piazza Diaz 2 20123 Milan Italy

Attention: [***]
Email: [***]

To Seller:

Liminal BioSciences Inc. 440, boul. Armand-Frappier Bureau 300 Laval, Québec H7V 4B4

Attention: [***] Email: [***]

with a copy to (which will not constitute notice):

Stikeman Elliott LLP 1155 Boulevard René-Lévesque West 41st Floor Montréal, Quebec H3B 3V2

Attention: [***] Email: [***]

or to such other street address, facsimile number, e-mail address or address as may be designated by notice given by any Party to the others. Any demand, notice or other communication given by personal delivery will be conclusively deemed to have been given on the day of actual delivery thereof and, if given by facsimile or e-mail, on the day of transmittal thereof if given during the normal business hours of the recipient and on the Business Day during which such normal business hours next occur if not given during such hours on any day.

15.10 No Third Party Beneficiaries.

Except as otherwise provided in Sections 12.1 and 12.2, the Parties intend that this Agreement shall not benefit or create any right or cause of action in, or on behalf of, any Person other than the Parties to this Agreement and no Person, other than the Parties to this Agreement shall be entitled to rely on the provisions of this Agreement in any action, suit, proceeding, hearing or other forum.

15.11 Governing Law and Attornment.

This Agreement will be governed by and construed in accordance with the Laws of the Province of Ontario and the Laws of Canada applicable therein, without giving effect to any choice of Law or conflict of Law provision or rule that would cause the application of the Laws of any jurisdiction other than the Province of Ontario. The Parties each irrevocably submit to the non-exclusive jurisdiction of the courts of the Province of Ontario in respect of any action or proceeding relating in any way to this Agreement (except with respect to disputes that are to be referred to the Independent Accountant pursuant to Section 3.3) and waive objection to the venue of any proceeding in such court or that such court provides an inconvenient forum.

113422456

15.12 Non-Merger.

Except as otherwise expressly provided in this Agreement, the covenants, representations and warranties shall not merge on and shall survive the Closing and, notwithstanding such Closing, shall continue in full force and effect.

15.13 Counterparts.

This Agreement may be executed in any number of counterparts (including counterparts by facsimile) and all such counterparts taken together shall be deemed to constitute one and the same instrument.

15.14 Severability.

If any provision of this Agreement shall be determined by an arbitrator or any court of competent jurisdiction to be illegal, invalid or unenforceable, that provision shall be severed from this Agreement and the remaining provisions shall continue in full force and effect.

15.15 Kedrion Code of Conduct.

Kedrion has implemented and put into effect a model for organization conduct, and control according to the Italian Law "D.Lgs.N. 231/2001" (the "Model") and implements a Code of Ethics Conduct which is part of the same (the "Italian Code"). The Model and the Italian Code are available at www.kedrion.it Seller shall report to Kedrion's surveillance body any breach by Seller or the Corporations prior to Closing. Purchaser shall inform Kedrion about and to its knowledge pending proceedings and/or censure of Seller or the Corporations for the crimes provided by D.Lgs. 231/01 relating to this Agreement. The violation of the foregoing reporting requirements will constitute a serious breach of this Agreement, and in such event Purchaser will have the right to terminate the Agreement if the breach is not promptly cured. The termination will be by right, following a simple written communication, the exercise of which shall not be deemed a waiver of any other right or remedy provided by this Agreement, applicable Law, or available in equity.

15.16 Force Majeure.

Neither Party shall be liable in any matter for failure to perform its obligations hereunder (such obligation(s) being an obligation other than a payment obligation), in whole or in part, all or any part of this Agreement when such failure is due to a cause or circumstance beyond such Party's reasonable control, including, but not limited to government action, war, terrorism, fire, explosion, flood, strike, lockout, embargo, shortage of raw materials, temporary export discontinuity for domestic supply, pandemic of disease (including avian flu or any other pandemic disease which can affect egg production), act of God (a "Force Majeure Event"), provided that the party claiming Force Majeure has exerted commercially reasonable efforts to avoid or remedy such Force Majeure, then such failure shall be excused for the duration of such event and for such a time as is reasonable to enable to resume performance under this Agreement. Upon cessation of such Force Majeure Event, the affected Party shall promptly resume performance of its obligations hereunder.

[Signature page follows.]

113422456

IN WITNESS WHEREOF THE FA	rties have executi	ed this Share Purchase Agreement as of the date first above written.	
PURCHASER:			
	KEDRION S.P.A		
	Ву:	(s) Signed	
		Name: [***] Title: [***]	
113422456			

SELLER:

LIMINAL BIOSCIENCES INC.

By: (s) Patrick Sartore

Name: Patrick Sartore

Title: President

Exhibit "A"

Illustrative Calculation of Working Capital and Net Indebtedness

[Exhibit Omitted.]

Exhibit "B"

Plasma Specifications

[Exhibit Omitted.]

Exhibit "C"

Forms of Certificate of Analysis for the Plasma Inventory

[Exhibit Omitted.]